1960CCO34877
BUSH ROSS GARDNER WARREN & RUDY, P.A.

MAHLON H. BARLOW, III
JOHN R. BUBH
JAMES O. DAVIS, III
BAMUEL B. DOLCIMABGOLD
FATRICIA LABARTA DOUGLAS
LEE E. DRYMN*
RICHARD K. FUEYO
J. STEPHEN GARDNER
JOHN N. GIORDANO
JEFFREY P. GREENBERG
RICHARD B. HADLOW
LIDA A. HOPPE

*Admitted only in Tennesses

PAUL L. HUEY

ATTORNEYS AT LAW 280 BOUTH PRANKLIN STREET TAMPA, PLONIDA 20002

(813) 224-9255

TELECOPIEN (815) 225-9520

March 12, 1996

DAVID M, JEFFRIER CHRISTINE M. POLANS ALEXANDRA M. RENARD JEREMY P. 1098 JOHN F. RIDY, II EDWARD O. SAVITZ ALICIA J. SCHUMACHER DANIEL H. BHERMAN, IV NEAL A. SIVYER H. BRADLEY STAGGS RANDY K. STERNS JEFFREY W. WARNEN PAUL D. WATSON DAVID S. WILLIAMS

Corporate Records Bureau Division of Corporations Department of State P.O. Box 6327 Tallahassee, Florida 32314

MAR 20 1996 354

Re:

FORT BROOKE BANCORPORATION

Our File No. FBSB-11

Gentlemen:

900001745939 -03/18/96--01013--001 ****122.50 ****122.50

On behalf of our captioned client, I am forwarding an original and one copy of its Articles of Incorporation, together with our firm check in the amount of \$122.50 in payment of the following charges:

1. Fee for filing Articles of Incorporation

2. Fee for obtaining certified copy of Articles of Incorporation

3. Filing a certificate designating Registered Agent

EFFECTIVE DATE
3-14-916

\$35.005 \$2.505 \$35.00 \$35.00 \$152.500

I would appreciate having you file the original Articles of Incorporation and the certificate designating Registered Agent and return to me a certified copy of the Articles as filed.

In connection with your action on such request, your records will reveal that there presently exists a Florida banking corporation using the name, Fort Brooke Bank. As counsel for that entity (as well as for the filing party), I represent that it consents to the use of the name, "Fort Brooke Bancorporation," by the filing party. Once incorporated, Fort Brooke Bancorporation will file an application with the Federal Reserve Bank of Atlanta for the purpose of qualifying as a bank holding company whose shares are to be issued to the present shareholders of Fort Brooke Bank following their approval of a merger transaction and the separate approval to such action by the Florida Department of Banking and Finance.

Le Blief

Corporate Records Bureau March 12, 1996 Page 2

Thank you for your help in this matter.

. \

Jeremy P. Ross

Yours truly,

JPR/bar Enclosures 92939.01

ARTICLES OF INCORPORATION FORT BROOKE BANCORPORATION

FILED 96 MAR 15 PM 2:42

The undersigned, acting as incorporator of the captioned corporation under the Topida RIDA Business Corporation Act, adopts the following Articles of Incorporation: EFFECTIVE DAIE 3-14-96

ARTICLE I Corporate Name and Principal Office

The name of this corporation is FORT BROOKE BANCORPORATION and its principal office and mailing address is 510 Vonderburg Drive, Brandon FL 33511.

ARTICLE II Commencement of Corporate Existence

The corporation shall come into existence on March 14, 1996.

ARTICLE III General Nature of Business

The corporation is formed for the purpose of qualifying and thereafter operating as a bank holding company, and of engaging in such related and permissible business activities in connection therewith as the corporation's board of directors may specify by resolution.

ARTICLE IV Capital Stock

Section 4.1 Capitalization. The aggregate number of shares of capital stock authorized to be issued by the Corporation shall be 10,000,000 shares of common stock, each with a par value of \$.001 (the "Common Stock"), and 2,000,000 shares of preferred stock, each with a par value of \$.001 (the "Preferred Stock"). Each share of issued and outstanding Common Stock shall entitle the holder thereof to fully participate in all shareholder meetings, to cast one vote on each matter with respect to which shareholders have the right to vote, and to share ratably in all dividends and other distributions declared and paid with respect to the Common Stock, as well as in the net assets of the corporation upon liquidation or dissolution, but each such share shall be subject to the rights and preferences of the Preferred Stock as hereinafter set forth.

Section 4.2 Issuance of Series of Preferred Stock. The Preferred Stock may be issued from time to time in one or more series in any manner permitted by law, as determined from time to time by the Board of Directors and stated in any resolution providing for the issuance of such shares adopted by the Board of Directors pursuant to authority hereby vested in it, each series to be appropriately designated, prior to the issuance of any shares thereof, by some distinguishing letter, number or title. All shares of each series of Preferred Stock shall be alike in every particular and of equal rank, have the same powers, preferences and rights and be subject to the same qualifications, limitations and restrictions, without distinction between the shares of different series thereof, except in regard to the following particulars, which may differ as to different series:

- (a) the periodic or other rate of dividends payable and the dates from which such dividends shall commence to accrue, if at all;
- (b) the manner in which, if at all, shares of a particular series may be redeemed and the amount payable upon a share redemption;
- (c) the amount payable upon any voluntary or involuntary liquidation, dissolution or winding up of the corporation;
- (d) the provisions of any sinking fund established with respect to the shares of a series;
- (e) the terms and rates of conversion or exchange, if shares of a series are convertible or exchangeable; and
 - (f) the provisions as to voting rights, if any, associated with shares of a series.

Before any shares of a particular series of Preferred Stock are issued, the designations of such series and its terms in respect of the foregoing particulars shall be fixed and determined by the Board of Directors in any manner permitted by law and stated in a resolution providing for the issuance of such shares adopted by the Board of Directors pursuant to authority hereby vested in it. Such designations and terms shall be set forth in full or summarized on the certificates for such series. The Board of Directors may increase the number of such shares by providing that any unissued shares of Preferred Stock shall constitute part of such series, or may decrease (but not below the number of shares thereof then outstanding) the number of shares of any series of Preferred Stock already created by providing that any unissued shares previously assigned to such series shall no longer constitute part thereof. The Board of Directors is hereby empowered to classify or reclassify any unissued shares of Preferred Stock by fixing or altering the terms thereof in respect of the above-referenced particulars and by assigning the same to an existing or newly established series from time to time before the issuance of such shares.

The holders of shares of each series shall be entitled to receive, out of any funds legally available therefor, when and as declared by the Board of Directors, cash dividends at such rate per annum as shall be fixed by resolution of the Board of Directors for such series, payable periodically on the dates fixed by the Board of Directors for the series. Such dividends may be cumulative or non-cumulative, deemed to accrue from day to day regardless of whether or not earned or declared, and may commence to accrue on each share of Preferred Stock from such date or dates, all as may be determined and stated by the Board of Directors prior to the issuance thereof. The corporation shall make dividend payments ratably upon all outstanding shares of Preferred Stock in proportion to the amount of dividends accrued thereon to the date of such dividend payment, if any.

As long as any shares of Preferred Stock shall remain outstanding, no dividend (other than a dividend payable in shares ranking junior to such Preferred Stock with respect to the payment of dividends or liquidating assets) shall be declared or paid upon, nor shall any distribution be made or ordered in respect of, shares of the Common Stock or any other class of shares ranking junior to the shares of such Preferred Stock as to the payment of dividends or liquidating assets, nor shall any monles (other than the net proceeds received from the sale of shares ranking junior to the shares of such Preferred Stock as to the payment of dividends or liquidating assets) be set aside for or applied to the purchase or redemption (through a sinking fund or otherwise) of shares of the Common Stock or of any other class of shares ranking junior to the shares of such Preferred Stock as to dividends or assets unless:

- (a) all dividends accrued with respect to the shares of Preferred Stock of all series for past dividend periods shall have been paid and the full dividend on all outstanding shares of Preferred Stock of all series for the then current dividend period shall have been paid or declared and set apart for payment; and
- (b) the corporation shall have set aside all amounts, if any, required to be set aside as and for sinking funds, if any, for the shares of Preferred Stock of all series for the then current year, and all defaults, if any, in complying with any such sinking fund requirements in respect of previous years shall have been cured.

The corporation, at the option of the Board of Directors, may at any time redeem the whole, or from time to time any part, of any series of Preferred Stock, subject to such limitations as may be adopted by the Board authorizing the issuance of such shares, by paying therefor in cash the amount which shall have been determined by the Board of Directors, in the resolution authorizing such series, to be payable upon the redemption of such shares at such time. Redemption may be made of the whole or any part of the outstanding shares of any one or more series, in the discretion of the Board of Directors; but if the redemption shall be effected only with respect to a part of a series, the shares to be redeemed may be selected by lot, or all of the shares of such series may be redeemed pro rata, in such manner as may be prescribed by resolution of the Board of Directors.

Subject to the foregoing provisions and to any qualifications, limitations or restrictions applicable to any particular series of Preferred Stock which may be stated in the resolution providing for the issuance of such series, the Board of Directors shall have authority to prescribe from time to time the manner in which any series of Preferred Stock shall be redeemed.

Upon any liquidation, dissolution or winding up of the corporation, whether voluntary or involuntary, the shares of Preferred Stock of each series shall be entitled, before any distribution shall be made with respect to shares of Common Stock or to any other class of shares junior to the shares of Preferred Stock as to the payment of dividends or liquidating assets, to be paid the full preferential amount fixed by the Board of Directors for such series as herein authorized; and thereafter shall be entitled to such further payment, if any, as shall be specified in the Board of Director resolution establishing the series. If upon such liquidation or

dissolution of the corporation, whether voluntary or involuntary, the net assets of the corporation shall be insufficient to permit the payment to all outstanding shares of Preferred Stock of all series of the full preferential amounts to which they are respectively entitled, the entire net assets of the corporation shall be distributed, in the order of seniority, fully as to each series with respect to which there are adequate net assets to satisfy the preferential amount and, as to the most senior series with respect to which there are inadequate net assets, ratably in proportion to the full preferential amount to which each share of that series is entitled. Neither a consolidation nor a merger of the corporation with or into any other entity nor the sale of all or substantially all of the assets of the corporation shall be deemed to be a liquidation or dissolution within the meaning of this paragraph.

ARTICLE V Initial Registered Office and Agent

The street address of the initial registered office of the corporation shall be 220 South Franklin Street, Tampa, Florida 33602, and the initial registered agent of the corporation at such address is Jeremy P. Ross.

ARTICLE VI Incorporator

The name and address of the corporation's incorporator is:

Name

Address

Barbara A. Rowe

220 South Franklin Street Tampa, Florida 33602

ARTICLE VII Directors

7.1 The Board of Directors shall be divided into three separate classes, designated Class I, Class II and Class III, each of which shall be composed of as nearly equal a number of directors as possible, with the initial membership of each Class to contain the individuals identified in Section 7.2 below. Each director first assigned to Class I shall serve for an initial term of one year, each director first assigned to Class II for an initial term of two years, and each director first assigned to Class III for an initial term of three years, with each director, irrespective of Class, being entitled to serve until the election and qualification of his or her successor or until his or her earlier resignation, death or removal from office. Upon the expiration of the initial term of office for each director, his or her successor (which may include the former director) shall be elected for a term of three years, to serve until the election and qualification of his or her successor or until an earlier resignation, death or removal from office.

7.2 The initial Board of Directors of the corporation shall consist of 14 members, whose identities, residential addresses and director Classes to which assigned are:

Name	Address	Director Class
John D. Adams,	7209 Adamo Drive Tampa, FL 33619	1
Melvin R. Belisle	505 Lakeside Drive Seffner, FL 33584	1
Tommy W. Brown	3010 N. 38th Street Tampa, FL 33605	1
Richard H. Eatman	607 Royal Crest Drive Brandon, FL 33511	1
H. Rex Etheredge	918 Centerbrook Drive Brandon, FL 33511	I
Joseph Garcia	2014 Woodberry Road Brandon, FL 33511	II
Riley L. Hogan	1704 Cottage Way Court Brandon, FL 33511	11
Alex D. MacKinnon	705 Centerbrook Drive Brandon, FL 33511	II
Thomas H. Miller	616 West Brandon Blvd. Brandon, FL 33511	II
Charles W. Poe	4601 San Miguel Tampa, FL 33629	II
William F. Poe	70 Ladoga Tampa, FL 33606	Ш
Reese T. Poppell	2807 Wedgewood Drive Plant City, FL 33567	Ш
Sam Rampello	1107 Riverhills Drive Temple Terrace, FL 33617	III

Q

Ш

7.3 Unless two-thirds of all directors then in office shall approve the proposed change, this Article VII may be amended or rescinded only by the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote in an election of directors, at any regular or special meeting of shareholders, and notice of the proposed change must be contained in the notice of the meeting.

ARTICLE VIII Bylaws

Except as provided in the succeeding sentence, the Board of Directors shall have the right to adopt, amend or repeal the Bylaws of the corporation by the affirmative vote of a majority of all directors then in office, and the shareholders shall have such right by the affirmative vote of a majority of the issued and outstanding shares of the corporation entitled to vote in an election of directors. Notwithstanding the foregoing authority, any amendment of the Bylaws of the corporation changing the number of directors shall require the affirmative vote of two-thirds of all directors then in office or the affirmative vote of the holders of two-thirds of the issued and outstanding shares of capital stock of the corporation entitled to vote thereon, at any regular or special meeting of the shareholders, and notice of the proposed change must be contained in the notice of such meeting.

ARTICLE IX Removal of Directors

- 9.1 At any shareholders' meeting with respect to which notice of such purpose has been given, the entire Board of Directors or any individual director may be removed without cause only by the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote in an election of directors.
- 9.2 At any meeting of the Board of Directors, with respect to which notice of such purpose has been given, any individual director may be removed without cause only by the affirmative vote of at least two-thirds of the directors of the corporation then in office.
- 9.3 At any shareholders' meeting with respect to which notice of such purpose has been given, the entire Board of Directors or any individual director may be removed with cause by the affirmative vote of the holders of a majority of the issued and outstanding shares of the corporation entitled to vote in an election of directors.
- 9.4 For purposes of this Article IX, a director of the corporation may be removed for cause only if (i) the director has been convicted of a felony; (ii) any bank regulatory authority having jurisdiction over the corporation requests or demands the removal; or (iii) at least two-

thirds of the directors of the corporation then in office, excluding the director to be removed, determine that the director's conduct has been inimical to the best interests of the corporation.

9.5 Unless two-thirds of the directors then in office shall approve the proposed change, this Article IX may be amended or rescinded only the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote in an election of directors, at any regular or special meeting of the shareholders, and notice of the proposed change must be contained in the notice of the meeting.

ARTICLE X Limitation of Liability

- 10.1 A director of the corporation is not personally liable for monetary damages to the corporation or any other person for any statement, vote, decision, or failure to act, regarding corporate management or policy, by a director, unless:
 - a. The director shall have breached or failed to perform any duty as a director; and
 - b. The director's breach of, or failure to perform, any duty constitutes:
 - (1) A violation of the criminal law, unless the director had reasonable cause to believe his or her conduct was lawful or had no reasonable cause to believe his conduct was unlawful;
 - (2) A transaction from which the director derived an improper material tangible personal benefit either directly or indirectly:
 - (3) A circumstance under which the liability provisions of Section 607.0834 of the Florida Business Corporation Act pertaining to the distribution of corporate assets are applicable;
 - (4) In a proceeding by or in the right of the corporation to procure a judgment in its favor or by or in the right of a shareholder, conscious disregard for the best interest of the corporation or willful misconduct; or
 - (5) In a proceeding by or in the right of someone other than the corporation or a shareholder, recklessness or an act or omission which was committed in bad faith or with malicious purpose or in a manner exhibiting wanton or willful disregard of human rights, safety, or property.

- 10.2 Any repeal or modification of this Article X by the shareholders of the corporation shall be prospective only and shall not adversely affect any right or protection of a director of the corporation existing at the time of such repeal or modification.
- 10.3 Unless two-thirds of the directors then in office shall approve the proposed change, this Article X may be amended or rescinded only with the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote thereon, at any regular or special meeting of the shareholders, and notice of the proposed change must be contained in the notice of the meeting.

ARTICLE XI SuperMajority Voting Requirements

- 11.1 Except as set forth in Section 11.4 of this Article XI, the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote thereon shall be required to approve:
 - a. any merger or share exchange of the corporation with or into any other corporation; or
 - b. any sale, lease, exchange or other disposition of all or substantially all of the assets of the corporation to any other corporation, person or other entity;

if, as of the record date for determination of shareholders entitled to notice thereof and to vote thereon, such other corporation, person or entity which is a party to such a transaction is the beneficial owner, directly or indirectly, of five percent (5%) or more of the issued and outstanding shares of the corporation entitled to vote in an election of directors.

- 11.2 For purposes of this Article XI, any corporation, person or other entity shall be deemed to be the beneficial owner of any shares of the corporation:
 - a. which it owns directly, whether or not of record; or
 - b. which it has the right to acquire, pursuant to any agreement or understanding or upon exercise of conversion rights, exchange rights, warrants or options or otherwise or the right to vote pursuant to any agreement, arrangement or understanding; or
 - c. which are beneficially owned, directly or indirectly (including shares deemed to be owned through application of subparagraph (b)(ii) above), by an "affiliate" or "associate" (as those terms are defined in Rule 12b-2 of the General Rules and Regulations under the Securities Exchange Act

- of 1934 as in effect on January 1, 1996 of the other corporation, person or entity; or
- d. which are beneficially owned, directly or indirectly (including shares deemed owned through application of subparagraph (b)(ii) above), by any other corporation, person or entity with which it or its "affiliate" or "associate" (as defined above) has any agreement or arrangement or understanding for the purpose of acquiring, holding, voting or disposing of shares of the corporation.

For the purpose of determining whether a corporation, person or entity is the beneficial owner of one or more of the issued and outstanding shares of the corporation, the issued and outstanding shares of the corporation shall include shares not in fact issued and outstanding but deemed owned through the application of clauses b., c. and d. above, but shall not include any other shares which are not then issued and outstanding but which may be issuable pursuant to any agreement or upon exercise of conversion rights, exchange rights, warrants, options or otherwise.

- 11.3 The Board of Directors shall have the power and duty to determine for the purposes of this Article XI, on the basis of information known to the corporation, whether:
 - a. such other corporation, person or entity beneficially owns, directly or indirectly, more than five percent (5%) of the issued and outstanding shares of the corporation entitled to vote in an election of directors;
 - b. a corporation, person or entity is an "affiliate" or "associate" (as defined above) of another:
 - c. any sale, lease, exchange or other disposition of part of the assets of the corporation involves substantially all of the assets of the corporation; and
 - d. the memorandum of understanding referred to in Section 11.4 below is substantially consistent with the transaction covered thereby.

Any such determination shall be conclusive and binding for all purposes of this Article

11.4 The provisions of this Article XI shall not apply to:

XI.

a. any merger or similar transaction with any corporation if two-thirds of all directors of the corporation then in office shall have approved a memorandum of understanding with such other corporation with respect to such other transaction prior to the time that such other corporation shall have become the beneficial owner of more than five percent (5%) of the

- issued and outstanding shares of the corporation entitled to vote in an election of directors; or, after such acquisition of 5% of the issued and outstanding shares, if two-thirds or more of the directors then holding office approve such transaction prior to its consummation; or
- b. any merger or share exchange of the corporation with, or any sale or lease to the corporation (or any subsidiary thereof) of any assets of, or any sale or lease by the corporation (or any subsidiary thereof) of any of its assets to, any corporation of which a majority of the outstanding shares of all classes of stock entitled to vote in an election of directors is owned of record or beneficially by the corporation and its subsidiaries.
- 11.5 Unless two-thirds of the directors then in office shall approve the proposed change, this Article XI may be amended or rescinded only the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote thereon, at any regular or special meeting of the shareholders, and notice of the proposed change must be contained in the notice of the meeting.

ARTICLE XII Standard of Review

- 12.1 The Board of Directors, when evaluating an offer of another party (i) to make a tender offer or exchange offer for any equity security of the corporation, (ii) to merge or consolidate any other corporation with the corporation, or (iii) to purchase or otherwise acquire all or substantially all of the assets of the corporation, shall, in determining what is in the best interest of the corporation and its shareholders, give due consideration to all relevant factors, including without limitation: (A) the short-term and long-term social and economic effects on the employees, customers, shareholders and other constituents of the corporation and its subsidiaries, it being understood that any subsidiary bank of the corporation is charged with providing support to and being involved in the communities it serves; and (B) the consideration being offered by the other party in relation to the then-current value of the corporation in a freely negotiated transaction and in relation to the Board of Directors' then-estimate of the future value of the corporation as an independent entity.
- 12.2 Unless two-thirds of the directors then in office shall approve the proposed change, this Article XII may be amended or rescinded only by the affirmative vote of the holders of at least two-thirds of the issued and outstanding shares of the corporation entitled to vote thereon, at any regular or special meeting of the shareholders, and notice of the proposed change must be contained in the notice of the meeting.

ARTICLE XIII Waiver of Statutory Provisions

6

The corporation elects not to be governed by Section 607.0901, Florida Statutes, relating to certain affiliated transactions, and separately elects that Section 607.0902, Florida Statutes, shall not apply to control-share acquisitions of the corporation's shares.

ARTICLE XIV Severability

Should any provision of these Articles of Incorporation, or any clause hereof, be held to be invalid, illegal or unenforceable, in whole or in part, the remaining provisions and clauses of these Articles of Incorporation shall remain valid and fully enforceable.

ARTICLE XV Indemnification

If in the judgment of a majority of the entire Board of Directors, (excluding from such majority any director under consideration for indemnification), the criteria set forth in §607.0850(1) or (2), Florida Statutes, as then in effect, have been met, then the corporation shall indemnify any director, officer, employee or agent thereof, whether current or former, together with his or her personal representatives, devisees or heirs, in the manner and to the extent contemplated by §607.0850, as then in effect, or by any successor law thereto.

IN WITNESS WHEREOF, the undersigned has executed these Articles this 12th day of March 1996.

Bachara A. Rowe

90879.04

CERTIFICATE DESIGNATING REGISTERED AGENT

Pursuant to the provisions of \$\$48.091 and 607.0501, Florida Statutes, FORT BROOKE BANCORPORATION, desiring to organize under the laws of the State of Florida, hereby designates Jeremy P. Ross, an individual resident of the State of Florida, as its Registered Agent for the purpose of accepting service of process within such State and designates 220 South Franklin Street, Tampa, Florida 33602, the business office of its Registered Agent, as its Registered Office.

FORT BROOKE BANCORPORATION

By Buckey M. Cource
Barbara A. Rowe, Incorporator

ACKNOWLEDGMENT

I hereby accept my appointment as Registered Agent of the above named corporation, acknowledge that I am familiar with and accept the obligations imposed by Florida law upon that position, and agree to act as such in accordance with the provisions of §§48.091 and 607.0505, Florida Statutes.

90879.04

P96000024877

MANLON H, RANLOW, III
JOHN N, BUSH
JAMES O, DAVIS, III
SAMUEL B, DDLCIMASCOLO
PATRICIA LABARTA DOUGLAS
LEE E, DRYER*
RICHARD K, FUSYO
J, STEPHEN GARDNER
JOHN N, GIORDANO
JEPPREY P, ORSENBERG
RICHARD B, HADLOW
LIBA A, HOPPE
PAUL L, HUEY

*Admitted only in Tennesses

ATTORNEYS AT LAW #30 BOUTH FRANKLIN STREET TAMPA, FLORIDA 33602

(013) 224-9200

TELECOPIEM (813) 223-8620

March 22, 1996

DAVID M. JEPPRIES
CHRISTINS M. POLANS
ALEXANDRA M. RENARD
JEREMY P. ROSS
JOHN F. RUDY, II
EDWARD O. SAVITZ
ALIGIA J. SCHUMACHER
DANIEL H. SHERMAN, IV
HALA. SIYYER
H. SRADLEY STAGGS
RANDY K. STPRNS
JEPPREY W. WARREN
PAUL D. WATBON
DAVID S. WILLIAMS

Corporate Records Bureau of Corporations Department of State P. O. Box 6327 Tailahassee, FL 32314

3000001757163 -03/26/96-01061-014 *****35.00 *****35.00

RE: Fort Brooke Bancorporation

Ladies/Gentlemen:

On behalf of the above captioned Corporation, I am forwarding an original and one copy of the its Articles of Amendment to and Restatement of the Articles of Incorporation as originally filed with your office as of March 14, 1996. Also enclosed is our firm check in the amount of \$35 in payment of the filing fee.

JPR/bar Enclosures
93882.01

AUTHORIZATION BY PHONE TO
CONNEUT ADDITE CLASSES

BATE 2/29

ARTICLES OF AMENDMENT TO AND RESTATEMENT OF THE ARTICLES OF INCORPORATION OF FORT BROOKE BANCORPORATION

FORT BROOKE BANCORPOFATION, a Florida corporation (the "Corporation"), hereby certifies as follows:

1. The Articles of Incorporation of the Corporation are hereby amended and ells so amended, are restated in their entirety to read as follows:

ARTICLES OF INCORPORATION OF FORT BROOKE BANCORPORATION

ARTICLE I Corporate Name and Principal Office

The name of the Corporation shall be Fort Brooke Bancorporation and the street address of its principal office, as of the date of these Articles of Incorporation, is 510 Vonderburg Drive, Brandon, Florida 33511.

ARTICLE II General Nature of Business

The Corporation is formed for the purpose of qualifying and thereafter operating as a bank holding company, and of engaging in such related and permissible business activities in connection therewith as the Corporation's board of directors may specify by resolution.

ARTICLE III Capital Stock

Section 3.1 <u>Capitalization</u>. The aggregate number of shares of capital stock authorized to be issued by the Corporation shall be 10,000,000 shares of common stock, each with a par value of \$.001 (the "Common Stock"), and 2,000,000 shares of preferred stock, each with a par value of \$.001 (the "Preferred Stock"). Each share of issued and outstanding Common Stock shall entitle the holder thereof to fully participate in all shareholder meetings, to cast one vote on each matter with respect to which shareholders have the right to vote, and to share ratably in all dividends and other distributions declared and paid with respect to the Common Stock, as well as in the net assets of the corporation upon liquidation or dissolution, but each such share shall be subject to the rights and preferences of the Preferred Stock as hereinafter set forth.

Section 3.2 <u>Issuance of Series of Preferred Stock</u>. The Preferred Stock may be issued from time to time in one or more series in any manner permitted by law, as determined from

time to time by the Board of Directors and stated in any resolution providing for the issuance of such shares adopted by the Board of Directors pursuant to authority hereby vested in it, each series to be appropriately designated, prior to the issuance of any shares thereof, by some distinguishing letter, number or title. All shares of each series of Preferred Stock shall be alike in every particular and of equal rank, have the same powers, preferences and rights and be subject to the same qualifications, limitations and restrictions, without distinction between the shares of different series thereof, except in regard to the following particulars, which may differ as to different series:

- (a) the periodic or other rate of dividends payable and the dates from which such dividends shall commence to accrue, if at all;
- (b) the manner in which, if at all, shares of a particular series may be redeemed and the amount payable upon a share redemption;
- (c) the amount payable upon any voluntary or involuntary liquidation, dissolution or winding up of the corporation;
- (d) the provisions of any sinking fund established with respect to the shares of a series:
- (e) the terms and rates of conversion or exchange, if shares of a series are convertible or exchangeable; and
 - (f) the provisions as to voting rights, if any, associated with shares of a series.

Before any shares of a particular series of Preferred Stock are issued, the designations of such series and its terms in respect of the foregoing particulars shall be fixed and determined by the Board of Directors in any manner permitted by law and stated in a resolution providing for the issuance of such shares adopted by the Board of Directors pursuant to authority hereby vested in it. Such designations and terms shall be set forth in full or summarized on the certificates for such series. The Board of Directors may increase the number of such shares by providing that any unissued shares of Preferred Stock shall constitute part of such series, or may decrease (but not below the number of shares thereof then outstanding) the number of shares of any series of Preferred Stock already created by providing that any unissued shares previously assigned to such series shall no longer constitute part thereof. The Board of Directors is hereby empowered to classify or reclassify any unissued shares of Preferred Stock by fixing or altering the terms thereof in respect of the above-referenced particulars and by assigning the same to an existing or newly established series from time to time before the issuance of such shares.

The holders of shares of each series shall be entitled to receive, out of any funds legally available therefor, when and as declared by the Board of Directors, cash dividends at such rate per annum as shall be fixed by resolution of the Board of Directors for such series, payable periodically on the dates fixed by the Board of Directors for the series. Such dividends may be

cumulative or non-cumulative, deemed to accrue from day to day regardless of whether or not carned or declared, and may commence to accrue on each share of Preferred Stock from such date or dates, all as may be determined and stated by the Board of Directors prior to the issuance thereof. The corporation shall make dividend payments ratably upon all outstanding shares of Preferred Stock in proportion to the amount of dividends accrued thereon to the date of such dividend payment, if any.

As long as any shares of Preferred Stock shall remain outstanding, no dividend (other than a dividend payable in shares ranking junior to such Preferred Stock with respect to the payment of dividends or liquidating assets) shall be declared or paid upon, nor shall any distribution be made or ordered in respect of, shares of the Common Stock or any other class of shares ranking junior to the shares of such Preferred Stock as to the payment of dividends or liquidating assets, nor shall any monies (other than the net proceeds received from the sale of shares ranking junior to the shares of such Preferred Stock as to the payment of dividends or liquidating assets) be set aside for or applied to the purchase or redemption (through a sinking fund or otherwise) of shares of the Common Stock or of any other class of shares ranking junior to the shares of such Preferred Stock as to dividends or assets unless:

- (a) all dividends accrued with respect to the shares of Preferred Stock of all series for past dividend periods shall have been paid and the full dividend on all outstanding shares of Preferred Stock of all series for the then current dividend period shall have been paid or declared and set apart for payment; and
- (b) the Corporation shall have set aside all amounts, if any, required to be set aside as and for sinking funds, if any, for the shares of Preferred Stock of all series for the then current year, and all defaults, if any, in complying with any such sinking fund requirements in respect of previous years shall have been cured.

The Corporation, at the option of the Board of Directors, may at any time redeem the whole, or from time to time any part, of any series of Preferred Stock, subject to such limitations as may be adopted by the Board authorizing the issuance of such shares, by paying therefor in cash the amount which shall have been determined by the Board of Directors, in the resolution authorizing such series, to be payable upon the redemption of such shares at such time. Redemption may be made of the whole or any part of the outstanding shares of any one or more series, in the discretion of the Board of Directors; but if the redemption shall be effected only with respect to a part of a series, the shares to be redeemed may be selected by lot, or all of the shares of such series may be redeemed pro rata, in such manner as may be prescribed by resolution of the Board of Directors.

Subject to the foregoing provisions and to any qualifications, limitations or restrictions applicable to any particular series of Preferred Stock which may be staud in the resolution providing for the issuance of such series, the Board of Directors shall have authority to prescribe from time to time the manner in which any series of Preferred Stock shall be redeemed.

Upon any liquidation, dissolution or winding up of the corporation, whether voluntary or involuntary, the shares of Preferred Stock of each series shall be entitled, before any distribution shall be made with respect to shares of Common Stock or to any other class of shares junior to the shares of Preferred Stock as to the payment of dividends or liquidating assets, to be paid the full preferential amount fixed by the Board of Directors for such series as herein authorized; and thereafter shall be entitled to such further payment, if any, as shall be specified in the Board of Director resolution establishing the series. If upon such liquidation or dissolution of the Corporation, whether voluntary or involuntary, the net assets of the Corporation shall be insufficient to permit the payment to all outstanding shares of Preferred Stock of all series of the full preferential amounts to which they are respectively entitled, the entire net assets of the Corporation shall be distributed, in the order of seniority, fully as to each series with respect to which there are adequate net assets to satisfy the preferential amount and, as to the most senior series with respect to which there are inadequate net assets, ratably in proportion to the full preferential amount to which each share of that series is entitled. Neither a consolidation nor a merger of the corporation with or into any other entity nor the sale of all or substantially all of the assets of the Corporation shall be deemed to be a liquidation or dissolution within the meaning of this paragraph.

ARTICLE IV Initial Registered Office and Agent

The street address of the initial registered office of the corporation shall be 220 South Franklin Street, Tampa, Florida 33602, and the initial registered agent of the corporation at such address is Jeremy P. Ross.

ARTICLE V Term of Existence

The term for which the Corporation shall exist shall be perpetual unless terminated pursuant to applicable Florida law.

ARTICLE VI Directors

- 6.1 The Board of Directors shall be composed of a membership consisting of no fewer than five nor more than 25 individuals, each of whom shall serve for a term of one year and until the election and qualification of his or her successor, subject to his or her earlier resignation or removal from office.
- 6.2 The initial Board of Directors of the corporation shall consist of 14 members, whose identities and residential addresses are:

Name	Address
John D. Adams,	7209 Adamo Drive Tampa, FL 33619
McIvin R. Belisle	505 Lakeside Drive Seffner, FL 33584
Tommy W. Brown	3010 N. 38th Street Tampa, FL 33605
Richard H. Eatman	607 Royal Crest Drive Brandon, FL 33511
H. Rex Etheredge	918 Centerbrook Drive Brandon, FL 33511
Joseph Garcia	2014 Woodberry Road Brandon, FL 33511
Riley L. Hogan	1704 Cottage Way Court Brandon, FL 33511
Alex D. MacKinnon	705 Centerbrook Drive Brandon, FL 33511
Thomas H. Miller	616 West Brandon Blvd. Brandon, FL 33511
Charles W. Poe	4601 San Miguel Tampa, FL 33629
William F. Poe	70 Ladoga Tampa, FL 33606
Reese T. Poppell	2807 Wedgewood Drive Plant City, FL 33567
Sam Rampello	1107 Riverhills Drive Temple Terrace, FL 33617
David C. Worthington	908 South Parsons Brandon, FL 33511

- 2. The foregoing amendment and restatement shall become effective as of the close of business on the date these Articles of Amendment and Restatement are approved by the Florida Department of State and all filing fees then due have been paid, all in accordance with the corporation laws of the State of Florida.
- 3. The amendment and restatement recited in Section 1, above has been duly adopted in accordance with the provisions of §§607,1003, .1007 and .0704, Florida Statutes, the Board of Directors of Fort Brooke Bancorporation having adopted a resolution setting forth such amendment, declaring its advisability and the Corporation's sole shareholder having approved of such amendment and restatement by executed written action. The date of adoption of the amendment is March 22, 1996.

IN WITNESS WHEREOF, Fort Brooke Bancorporation has caused these Articles of Amendment and Restatement to be prepared under the signature of its President and the attestation of its Secretary this 22nd day of March 1996.

FORT BROOKE BANCORPORATION

By: Richard H. Eatman, President

93876

96000024877 BUSH ROSS GARDNER WARREN & RUDY, P.A.

MARLON H. BARLOW, HI JOHN R. BUBH JAMES O, DAVIS, III BAMULL B. DOLCIMARCOLD PATRICIA LABARTA DOUGLAS

LEE E. DRYER! NICHAND K. PULYO J. BTEPHEN GANDNER JOHN N. GIORDANO JEPPREY P. GREENBERG RICHARD N. HADLOW LIBA A. HOPPE PAUL L. HURY *Admitted only in Tennesses

ATTORNEYS AT LAW 220 SOUTH FRANKLIN STREET EDBEE AUINGL ANNAT

(013) 224-9266

TELECOPIEH (813) 223-0020

April 5, 1996

DAVID M. JEFFRIER CHRISTINE M. POLANS ALEXANDRA M. HENARD JEREMY P. HORS JOHN F. RUDY, II EDWARD O. SAVITE ALICIA J. BOHUMACHER DANIEL H. BHERMAN, IV NEAL A. BIVYER H. BRADLEY STAGGS MANDY K. STERNB JEFFREY W. WARREN PAUL D. WATBON DAVID B. WILLIAMS

grounds Pressay:

- MZ99Z96 - D1069 - UUL

Corporate Records Bureau of Corporations Department of State P. O. Box 6327 Tallahassee, FL 32314

> RE: Fort Brooke Bancorporation

Ladies/Gentlemen:

On behalf of the above captioned Corporation, I am forwarding an original and one copy of its Articles of Amendment to and Restatement of the Articles of Incorporation as originally filed with your office as of March 14, 1996, and subsequently amended and restated as of March 25. Also enclosed is our firm check in the amount of \$88.50 for payment of the filing fee and a certified copy.

Sincerely,

Jeremy P. Ross

L APR 1 1 1996

JPR/bar Enclosures 94923.01

ARTICLES OF AMENDMENT TO AND RESTATEMENT OF THE ARTICLES OF INCORPORATION OF FORT BROOKE BANCORPORATION

FORT BROOKE BANCORPORATION, a Florida corporation (the "Corporation"), hereby certifies as follows:

1. The Articles of Incorporation of the Corporation are hereby amended and, as so amended, are restated in their entirety to read as follows:

ARTICLES OF INCORPORATION OF FORT BROOKE BANCORPORATION

ARTICLE I Corporate Name and Principal Office

The name of the Corporation shall be Fort Brooke Bancorporation and the street address of its principal office, as of the date of these Articles of Incorporation, is 510 Vonderburg Drive, Brandon, Florida 33511.

ARTICLE II General Nature of Business

The Corporation is formed for the purpose of qualifying and thereafter operating as a bank holding company, and of engaging in such related and permissible business activities in connection therewith as the Corporation's board of directors may specify by resolution.

ARTICLE III Capital Stock

Section 3.1 <u>Capitalization</u>. The aggregate number of shares of capital stock authorized to be issued by the Corporation shall be 10,000,000 shares of common stock, each with a par value of \$.001 (the "Common Stock"). Each share of issued and outstanding Common Stock shall entitle the holder thereof to fully participate in all shareholder meetings, to cast one vote on each matter with respect to which shareholders have the right to vote, and to share ratably in all dividends and other distributions declared and paid with respect to the Common Stock, as well as in the net assets of the corporation upon liquidation or dissolution.

ARTICLE IV Initial Registered Office and Agent

The street address of the initial registered office of the Corporation shall be 220 South Franklin Street, Tampa, Florida 33602, and the initial registered agent of the Corporation at such address is Jeremy P. Ross.

ARTICLE V Term of Existence

The term for which the Corporation shall exist shall be perpetual unless terminated pursuant to applicable Florida law.

ARTICLE VI Directors

- 6.1 The Board of Directors shall be composed of a membership consisting of no fewer than five nor more than 25 individuals, each of whom shall serve for a term of one year and until the election and qualification of his or her successor, subject to his or her earlier resignation or removal from office.
- 6.2 The initial Board of Directors of the corporation shall consist of 14 members, whose identities and residential addresses are:

Name	Address
John D. Adams,	7209 Adamo Drive Tampa, FL 33619
Melvin R. Belisle	505 Lakeside Drive Seffner, FL 33584
Tommy W. Brown	3010 N. 38th Street Tampa, FL 33605
Richard H. Eatman	607 Royal Crest Drive Brandon, FL 33511
H. Rex Etheredge	918 Centerbrook Drive Brandon, FL 33511
Joseph Garcia	2014 Woodberry Road Brandon, FL 33511

Riley L. Hogan

1704 Cottage Way Court

Brandon, FL 33511

Alex D. MacKinnon

705 Centerbrook Drive Brandon, FL 33511

Thomas H. Miller

616 West Brandon Blvd. Brandon, FL 33511

Charles W. Poc

4601 San Miguel Tampa, FL 33629

William F. Poe

70 Ladoga

Tampa, FL 33606

Reese T. Poppell

2807 Wedgewood Drive Plant City, FL 33567

Sam Rampello

1107 Riverhills Drive Temple Terrace, FL 33617

David C. Worthington

908 South Parsons Brandon, FL 33511

ARTICLE VII Waiver of Statutory Provisions

The Corporation elects not to be governed by Section 607.0901, Florida Statutes, relating to certain affiliated transactions, and separately elects that Section 607.0902, Florida Statutes, shall not apply to control-share acquisitions of the corporation's shares.

.

- 2. The foregoing amendment and restatement shall become effective as of the close of business on the date these Articles of Amendment and Restatement are approved by the Florida Department of State and all filing fees then due have been paid, all in accordance with the corporation laws of the State of Florida.
- 3. The amendment and restatement recited in Section 1. above has been duly adopted in accordance with the provisions of §§607.1003, .1007 and .0704, Florida Statutes, the Board of Directors of Fort Brooke Bancorporation having adopted a resolution setting forth such amendment, declaring its advisability and the Corporation's sole shareholder having approved of such amendment and restatement by executed written action, all as of April 1, 1996.

IN WITNESS WHEREOF, Fort Brooke Bancorporation has caused these Articles of Amendment and Restatement to be prepared under the signature of its President and the attestation of its Secretary this 4th day of April 1996.

FORT BROOKE BANCORPORATION

Richard H. Eatman, President

94519.01

ARTICLES OF INCORPORATION OF FORT BROOKE BANCORPORATION

ARTICLE I Corporate Name and Principal Office

The name of the Corporation shall be Fort Brooke Bancorporation and the street address of its principal office, as of the dr.e of these Articles of Incorporation, is 510 Vonderburg Drive, Brandon, Florida 33511.

ARTICLE II General Nature of Business

The Corporation is formed for the purpose of qualifying and thereafter operating as a bank holding company, and of engaging in such related and permissible business activities in connection therewith as the Corporation's board of directors may specify by resolution.

ARTICLE III Capital Stock

Section 3.1 <u>Capitalization</u>. The aggregate number of shares of capital stock authorized to be issued by the Corporation shall be 10,000,000 shares of common stock, each with a par value of \$.001 (the "Common Stock"). Each share of issued and outstanding Common Stock shall entitle the holder thereof to fully participate in all shareholder meetings, to cast one vote on each matter with respect to which shareholders have the right to vote, and to share ratably in all dividends and other distributions declared and paid with respect to the Common Stock, as well as in the net assets of the corporation upon liquidation or dissolution.

ARTICLE IV Initial Registered Office and Agent

The street address of the initial registered office of the Corporation shall be 220 South Franklin Street, Tampa, Florida 33602, and the initial registered agent of the Corporation at such address is Jeremy P. Ross.

ARTICLE V Term of Existence

The term for which the Corporation shall exist shall be perpetual unless terminated pursuant to applicable Florida law.

ARTICLE VI Directors

- 6.1 The Board of Directors shall be composed of a membership consisting of no fewer than five nor more than 25 individuals, each of whom shall serve for a term of one year and until the election and qualification of his or her successor, subject to his or her earlier resignation or removal from office.
- 6.2 The initial Board of Directors of the corporation shall consist of 14 members, whose identities and residential addresses are:

Name	Address
John D. Adams,	7209 Adamo Drive Tampa, FL 33619
Melvin R. Belisle	505 Lakeside Drive Seffner, FL 33584
Tominy W. Brown	3010 N. 38th Street Tampa, FL 33605
Richard H. Eatman	607 Royal Crest Drive Brandon, FL 33511
H. Rex Etheredge	918 Centerbrook Drive Br. ndon, FL 33511
Joseph Garcia	2014 Woodberry Road Brandon, FL 33511
Riley L. Hogan	1704 Cottage Way Court Brandon, FL 33511
Alex D. MacKinnon	705 Centerbrook Drive Brandon, FL 33511
Thomas H. Miller	' '6 West Brandon Blvd. Brandon, FL 33511
Charles W. Poe	4601 San Miguel Tampa, FL 37620
William F. Poe	70 Ladoga Tampa, FL 33606

Reese T. Poppett

2807 Wedgewood Drive

Plant City, FL 33567

Sam Rampello

1107 Riverhills Drive

Temple Terrace, FL 33617

David C. Worthington

908 South Parsons Brandon, FL 33511

ARTICLE VII Waiver of Statutory Provisions

The Corporation elects not to be governed by Section 607.0901, Florida Statutes, relating to certain affiliated transactions, and separately elects that Section 607.0902, Florida Statutes, shall not apply to control-share acquisitions of the corporation's shares.

93886.01

P960008487

		▼ • • • • • • • • • • • • • • • • • • •
C T CORPORATION SYST	EM	
lequestor's Nume 660 East Jefferson S	t-mart	
	rreet	
ddress Tallahassee, Florida	32301	CHTHTHTICA 1 CHTHTM: 1 FE
City State Zip	Phone ,	では「11「11」 - 14/22/27 - 11 1171 - 11 1日 - 14/22/27 - 11 1171 - 11日日 ******* 171 - 11 - まかままでり、ロロ
CORPORATIO	N(S) NAME	4. T. V. A. W. C. V.
ELD 4 D		M 0 m 0 /
tort Brooke Ban	corporation	10 starger
<u>and</u>	·	
The Colonial Bou	Maroup Tion	
() Profit		_
) NonProfit	() Amendment	⊘K) Merger
) Limited Liability Company		
() Foreign	() Dissolution/Witho	irawal () Mark
() Limited Partnership	() Annual Report	() Other
() Reinstatement	() Reservation	() Change of R.A.
() Limited Liability Partner	• • • • • • • • • • • • • • • • • • • •	() Fictitious_Name
() Certified Copy	() Photo Copies	() CUS
() Call When Ready	() Call if Problem	() After 4;30
(d∕Walk In	() Will Wait	Pick Up
() Mail Out		
Name		5 10
Availability /		PLEASE RETURN EXTRA COPY(S)
Document C		FILE STAMPED 2
Examiner	33 I	
Updater	4/22/97	<u> </u>
	. ()	<u> </u>
Verifier		
Acknowledgment		
W.P. Verifier		

CR2E031 (1-89)

P960000004877

ARTICLES OF MERGER Merger Sheet

MERGING:

FORT BROOKE BANCORPORATION, a Florida corporation P96000024877

INTO

THE COLONIAL BANCGROUP, INC., a Delaware corporation not qualified in Florida.

File date: April 23, 1997

Corporate Specialist: Annette Hogan

ARTICLES OF MERGER

OF

FORT BROOKE BANCORPORATION

AND

THE COLONIAL BANCGROUP, INC.

The undersigned corporations, FORT BROOKE BANCORPORATION and THE COLONIAL BANCGROUP, INC., file these Articles of Merger and certify that:

- 1. Fort Brooke Bancorporation, a Florida corporation, is hereby merged with and into The Colonial BancGroup, Inc., a Delaware corporation, pursuant to an Agreement and Plan of Merger dated as of November 18, 1996 (the "Plan of Merger"), a copy of which is attached as Exhibit A, and in accordance with the provisions of the Florida Business Corporation Act and the General Corporation Law of Delaware.
- 2. The surviving corporation is The Colonial BancGroup, Inc., a Delaware corporation.
- 3. The merger shall be effective at 11:00 a.m., Eastern Standard Time, on April 22, 1997.
- 4. The Plan of Merger was approved by the Board of Directors of The Colonial BancGroup, Inc., on October 16, 1996. Pursuant to the General Corporation Law of Delawarc, approval of the Plan of Merger by the shareholders of The Colonial BancGroup, Inc., was not required.
- 5. The Plan of Merger was approved by the Board of Directors of Fort Brooke Bancorporation on September 18, 1996, and was approved by the shareholders of Fort Brooke Bancorporation on March 13, 1997.

Dated: April 22, 1997.

FORT BROOKE BANCORPORATION

THE COLONIAL BANCGROUP, INC.

Richard H. Eatman

President and CEO

Robert E. Lowder

Chairman of the Board of Directors

President and CEO

z:\cbg.fbi\fl-art.mer

EXHIBIT A

AGREEMENT AND PLAN OF MERGER

by and between

THE COLONIAL BANCGROUP, INC.,

and

FORT BROOKE BANCORPORATION

dated as of

November 18, 1996

TABLE OF CONTENTS

		Page
	E I NAME	
1.1	Name	A-1
ARTICL	E 2 — MERGER — TERMS AND CONDITIONS	
2.1	Applicable Law	Λ-1
2.2	Corporate Existence	A-1
2.3	Articles of Incorporation and Bylaws	A-I
2.4	Resulting Corporation's Officers and Board	A-2
2.5	Stockholder Approval	A-2
2.6	Further Acts	A-2
2.7	Effective Date and Closing	Λ-2
2.8	Subsidiary Bank Merger	A-2
ADTICL	E 3—CONVERSION OF ACQUIRED CORPORATION STOCK	
3.1	Conversion of Acquired Corporation Stock	A-2
3.1	Surrender of Acquired Corporation Stock	A-3
3.3	Fractional Shares	A-4
3.3 3.4	Adjustments	A-4
3.4 3.5		A-4
	BancGroup Stock	A-4 A-4
3.6	Dissenting Rights	A-4
ARTICL	E 4—REPRESENTATIONS, WARRANTIES AND COVENANTS OF BANCGROUP	
4.1	Organization	A-4
4.2	Capital Stock	A-4
4.3	Financial Statements; Taxes	A-4
4.4	No Conflict with Other Instrument	A-5
4.5	Absence of Material Adverse Change	A-5
4.6	Approval of Agreements	A-5
4.7	Tax Treatment	A-6
4.8	Title and Related Matters	A-6
4.9	Subsidiaries	A-6
4.10	Contracts	A-6
4.11	Litigation	A-6
4.12	Compliance	A-6
4.13	Registration Statement	A-6
4.14	SEC Filings	A-7
4.15	Form S-4	A-7
4.16	Brokers	A-7
4.17	Government Authorization	A-7
4.18	Absence of Regulatory Communications	A-7
4.19	Disclosure	A-7
ARTICL	E 5—REPRESENTATIONS, WARRANTIES AND COVENANTS OF ACQUIRED CORPORATION	
5.1	Organization	A-8
5.2	Capital Stock	A-8
5.3	Subsidiaries	A-8
5.4	Financial Statements; Taxes	A-8
5.5	Absence of Certain Changes or Events	A-9
5.6	Title and Related Matters	A-10
5.7	Commitments	A-11

		Page
5.8	Charter and Bylaws	A-11
5.9	Litigation	A-11
5.10	Material Contract Defaults	A-11
5.11	No Conflict with Other Instrument	A-11
5.12	Governmental Authorization	A-11
5.13	Absence of Regulatory Communications	A-11
5.14	Absence of Material Adverse Change	A-12
5.15	Insurance	A-12
5.16	Pension and Employee Benefit Plans	A-12
5.17	Buy-Setl Agreement	A-12
5.18	Brokers	A-12
5.19	Approval of Agreements	A-12
5.20	Disclosure	A-13
5.21	Registration Statement	A-13
5.22	Loans; Adequacy of Allowance for Loan Losses	A-13
5.23	Environmental Matters	A-13
5.24	Transfer of Shares	A-13
5.25	Collective Bargaining	A-14
5.26	Labor Disputes	A-14
5.27	Derivative Contracts	A-14
		W-14
	E 6 — ADDITIONAL COVENANTS	
6.1	Additional Covenants of BancGroup	A-14
6.2	Additional Covenants of Acquired Corporation	A-16
ARTICL	E 7 — MUTUAL COVENANTS AND AGREEMENTS	
7.1	Best Efforts; Cooperation	A-18
7.2	Press Release	A-18
7.3	Mutual Disclosure	A-18
7.4	Access to Properties and Records	A-19
7.5	Notice of Adverse Changes	A-19
	E 8 — CONDITIONS TO OBLIGATIONS OF ALL PARTIES	
8.1	Approval by Shareholders	A-19
8.2	Regulatory Authority Approval	A-19
8.3	Litigation	A-19
8.4	Registration Statement	A-19
8.5	Tax Opinion	A-19
	•	M-20
	E 9 — CONDITIONS TO OBLIGATIONS OF ACQUIRED CORPORATION	
9.1	Representations, Warranties and Covenants	A-20
9.2	Adverse Changes	A-20
9.3	Closing Certificate	A-20
9.4	Opinion of Counsel	A-21
9.5	Fairness Opinion	A-21
9.6	NYSE Listing	A-21
9.7	Other Matters	A-21
9.8	Material Events	A-21
ARTICI	E 10 — CONDITIONS TO OBLIGATIONS OF BANCGROUP	
10.1	Representations, Warranties and Covenants	A-21
10.2	Adverse Changes	A-21
10.2	Closing Certificate	A-22
10.4	Opinion of Counsel	A-22
10.5	Controlling Shareholders	A-22

		Page
10.6	Other Matters	A-22
10.7	Dissenters	A-22
10.8	Material Events	A-2
10.9	Severance Agreements	Λ-2
10.10	Pooling of Interest	
ARTICL	E 11 — TERMINATION OF REPRESENTATIONS AND WARRANTIES	Λ-2
ARTICL	E 12 NOTICES	Λ-2
ARTICL	E 13 — AMENDMENT OR TEPMINATION	
13.1	- Amendment	
13.2	Termination	A-2
13.3	Damages	A-2
ARTICL	E 14 — DEFINITIONS	A-2
ARTICL	E 15 — MISCELLANEOUS	
15.1	Expenses	A-2
15.2	Benefit	A-2
15.3	Governing Law	A-2
15.4	Counterparts	A-2
15,5	Headings	A-2
15.6	Severability	A-2
15.7	Construction	
15.8	Return of Information	A-2
15.9	Equitable Remedies	
15.10	Attorneys' Fees	
15.11	No Waiver	
15.12	Remedies Cumulative	
15.13	Entire Contract	A-3

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER is made and entered into as of this the 18th day of November 1996, by and between FORT BROOKE BANCORPORATION ("Acquired Corporation"), a Florida corporation, and THE COLONIAL BANCGROUP, INC. ("BancGroup"), a Delaware corporation.

WITNESSETH

WHEREAS, Acquired Corporation operates as a bank holding company for its wholly owned subsidiary, Fort Brooke Bank (the "Bank"), with its principal office in Brandon, Florida; and

WHEREAS, BancGroup is a bank holding company with Subsidiary banks in Alabama, Florida, Georgia and Tennessee; and

WHEREAS, Acquired Corporation wishes to merge with BancGroup; and

WHEREAS, it is the intention of BancGroup and Acquired Corporation that such Merger shall qualify for federal income tax purposes as a "reorganization" within the meaning of section 368(a) of the Code, as defined herein:

NOW, THEREFORE, in consideration of the mutual covenants contained herein, the Parties hereto agree as follows:

ARTICLE 1

NAME

1.1 Name. The name of the corporation resulting from the Merger shall be "The Colonial BancGroup, Inc."

ARTICLE 2

MERGER - TERMS AND CONDITIONS

- 2.1 Applicable Law. On the Effective Date, Acquired Corporation shall be merged with and into BancGroup (herein referred to as the "Resulting Corporation" whenever reference is made to it as of the time of merger or thereafter). The Merger shall be undertaken pursuant to the provisions of and with the effect provided in the DGCL and, to the extent applicable, the FBCA. The offices and facilities of Acquired Corporation and of BancGroup shall become the offices and facilities of the Resulting Corporation.
- 2.2 Corporate Existence. On the Effective Date, the corporate existence of Acquired Corporation and of BancGroup shall, as provided in the DGCL and the FBCA, be merged into and continued in the Resulting Corporation, and the Resulting Corporation shall be deemed to be the same corporation as Acquired Corporation and BancGroup. All rights, franchises and interests of Acquired Corporation and BancGroup, respectively, in and to every type of property (real, personal and mixed) and choses in action shall be transferred to and vested in the Resulting Corporation by virtue of the Merger without any deed or other transfer. The Resulting Corporation on the Effective Date, and without any order or other action on the part of any court or otherwise, shall hold and enjoy all rights of property, franchises and interests, including appointments, designations and nominations and all other rights and interests as trustee, executor, administrator, transfer agent and registrar of stocks and bonds, guardian of estates, assignee, and receiver and in every other fiduciary capacity and in every agency, and capacity, in the same manner and to the same extent as such rights, franchises and interests were held or enjoyed by Acquired Corporation and BancGroup, respectively, on the Effective Date.
- 2.3 Articles of Incorporation and Bylaws. On the Effective Date, the certificate of incorporation and bylaws of the Resulting Corporation shall be the restated certificate of incorporation and bylaws of BancGroup as they exist immediately before the Effective Date.

- 2.4 Resulting Corporation's Officers and Board. The board of directors and the officers of the Resulting Corporation on the Effective Date shall consist of those persons serving in such capacities of BancGroup as of the Effective Date.
- 2.5 Stockholder Approval. This Agreement shall be submitted to the shareholders of Acquired Corporation at the Stockholders Meeting to be held as promptly as practicable consistent with the satisfaction of the conditions set forth in this Agreement. Upon approval by the requisite vote of the shareholders of Acquired Corporation as required by applicable Law, the Merger shall become effective as soon as practicable thereafter in the manner provided in section 2.7 hereof.
- 2.6 Further Acts. If, at any time after the Effective Date, the Resulting Corporation shall consider or be advised that any further assignments or assurances in law or any other acts are necessary or desirable (i) to vest, perfect, confirm or record, in the Resulting Corporation, title to and possession of any property or right of Acquired Corporation or BancGroup, acquired as a result of the Merger, or (ii) otherwise to carry out the purposes of this Agreement, BancGroup and its officers and directors shall execute and deliver all such proper deeds, assignments and assurances in law and do all acts necessary or proper to vest, perfect or confirm title to, and possession of, such property or rights in the Resulting Corporation and otherwise to carry out the purposes of this Agreement; and the proper officers and directors of the Resulting Corporation are fully authorized in the name of Acquired Corporation or BancGroup, or otherwise, to take any and all such action.
- 2.7 Effective Date and Closing. Subject to the terms of all requirements of Law and the conditions specified in this Agreement, the Merger shall become effective on the date specified in the Certificate of Merger to be issued by the Secretary of State of the State of Delaware (such time being herein called the "Effective Date"). The Closing shall take place at the offices of BancGroup, in Montgomery, Alabama, at 11:00 a.m. on the date that the Effective Date occurs or at such other place and time that the Parties may mutually agree.
- 2.8 Subsidiary Bank Merger. BancGroup and Acquired Corporation anticipate that immediately after the Effective Date the Bank will merge with and into Colonial Bank, BancGroup's Florida incorporated Subsidiary bank ("Colonial Bank"). The exact timing and structure of such merger are not known at this time, and BancGroup in its discretion will finalize such timing and structure at a later date. Acquired Corporation will cooperate with BancGroup in the execution of appropriate documentation relating to such merger.

ARTICLE 3

CONVERSION OF ACQUIRED CORPORATION STOCK

3.1 Conversion of Acquired Corporation Stock. (a) On the Effective Date, each share of common stock of Acquired Corporation outstanding and held by Acquired Corporation's shareholders (the "Acquired Corporation Stock"), shall be converted by operation of law and without any action by any holder thereof into shares of BancGroup Common Stock at a per share price of \$31.50 (the "Merger Consideration") as specified below. Specifically, each outstanding share of Acquired Corporation Stock shall (subject to section 3.3 hereof), be converted into such number of shares of BancGroup Common Stock equal to \$31.50 divided by the Market Value. The "Market Value" shall represent the per share market value of the BancGroup Common Stock at the Effective Date and shall be determined by calculating the average of the closing prices of the Common Stock of BancGroup as reported by the NYSE on each of the ten (10) consecutive trading days ending on the trading day immediately preceding the Effective Date, provided that if the Market Value, as calculated, is less than \$32.00, the Market Value shall be deemed to be \$32.00, and if the Market Value, as calculated, is greater than \$39.00, the Market Value shall be deemed to be \$39.00. Accordingly, the maximum number of shares of BancGroup Common Stock to be issued in the Merger shall be 975,076 and the minimum number of shares of BancGroup Common Stock to be issued in the Merger shall be 800,062 assuming 990,553 shares of Acquired Corporation common stock outstanding. To the extent that the number of shares of Acquired Corporation Stock may increase based upon the exercise of Acquired Corporation Options, the number of shares of BancGroup Common Stock to be issued in the Merger shall be increased with each share

of Acquired Corporation Stock outstanding at the Effective Date exchanged for shares of BaneGroup Common Stock equal to \$31.50 divided by the Market Value, provided that for this purpose the Market Value shall be deemed to be no less than \$32.00 and no greater than \$39.00.

- (b) (i) On the Effective Date, BancGroup shall assume all Acquired Corporation Options outstanding. and each such option shall cease to represent a right to acquire Acquired Corporation common stock and shall, instead, represent the right to acquire BancGroup Common Stock on substantially the same terms applicable to the Acquired Corporation Options except as specified below in this section. The number of shares of BancGroup Common Stock to be issued pursuant to such options shall equal the number of shares of Acquired Corporation common stock subject to such Acquired Corporation Options multiplied by the Exchange Ratio, provided that no fractions of shares of BancGroup Common Stock shall be issued and the number of shares of BancGroup Cemmon Stock to be issued upon the exercise of Acquired Corporation Options, if a fractional share exists, shall equal the number of whole shares obtained by rounding to the nearest whole number, giving account to such fraction. The exercise price for the acquisition of BancGroup Common Stock shall be the exercise price for each share of Acquired Corporation common stock subject to such options divided by the Exchange Ratio, adjusted appropriately for any rounding to whole shares that may be done. For purposes of this section 3.1(b) (i), the "Exchange Ratio" shall mean the result obtained by dividing \$31.50 by the Market Value. It is intended that the assumption by BancGroup of the Acquired Corporation Options shall be undertaken in a manner that will not constitute a "modification" as defined in Section 424 of the Code as to any stock option which is an "incentive stock option." Schedule 3.1 hereto sets forth the names of all persons holding Acquired Corporation Options, the number of shares of Acquired Corporation common stock subject to such options, the exercise price and the expiration date of such options.
- (ii) As an alternative to the assumption of Acquired Corporation Options provided for in section 3.1(b)(i), the holder of each such option may elect on the Effective Date, by properly surrendering to Acquired Corporation the Acquired Corporation Options, to receive shares of BancGroup Common Stock for the difference between the exercise price (the "Exercise Price") of the Acquired Corporation Options so surrendered and \$31.50. The number of shares of BancGroup Common Stock to be issued to such holder shall be such number of shares as shall equal (\$31.50 less the Exercise Price) divided by the Market Value.
- (iii) BancGroup shall file at its expense a registration statement with the SEC on Form S-8 or such other appropriate form (including the Form S-4 to be filed in connection with the Merger) with respect to the shares of BancGroup Common Stock to be issued pursuant to Acquired Corporation Options and shall use its reasonable best efforts to maintain the effectiveness of such registration statement for so long as such options remain outstanding. Such shares shall also be registered or qualified for sale under the securities laws of any state in which registration or qualification is necessary.
- 3.2 Surrender of Acquired Corporation Stock. After the Effective Date, each holder of an outstanding certificate or certificates which prior thereto represented shares of Acquired Corporation Stock who is entitled to receive BancGroup Common Stock shall be entitled, upon surrender to BancGroup of their certificate or certificates representing shares of Acquired Corporation Stock (or an affidavit or affirmation by such holder of the loss, theft, or destruction of such certificate or certificates in such form as BancGroup may reasonably require and, if BancGroup reasonably requires, a bond of indemnity in form and amount, and issued by such sureties, as BancGroup may reasonably require), to receive in exchange therefor a certificate or certificates representing the number of whole shares of BancGroup Common Stock into and for which the shares of Acquired Corporation Stock so surrendered shall have been converted, such certificates to be of such denominations and registered in such names as suc': holder may reasonably request. Until so surrendered and exchanged, each such outstanding certificate which, prior to the Effective Date, represented shares of Acquired Corporation Stock and which is to be converted into BancGroup Common Stock shall for all purposes evidence ownership of the BancGroup Common Stock into and for which such shares shall have been so converted, except that no dividends or other distributions with respect to such BancGroup Common Stock shall be made until the certificates previously representing shares of Acquired Corporation Stock shall have been properly tendered.

- 3.3 Fractional Shares. No fractional shares of BancGroup Common Stock shall be issued, and each holder of shares of Acquired Corporation Stock having a fractional interest arising upon the conversion of such shares into shares of BancGroup Common Stock shall, at the time of surrender of the certificates previously representing Acquired Corporation Stock, be paid by BancGroup an amount in cash equal to the Market Value of such fractional share.
- 3.4 Adjustments. In the event that prior to the Effective Date BancGroup Common Stock shall be changed into a different number of shares or a different class of shares by reason of any recapitalization or reclassification, stock dividend, combination, stock split, or reverse stock split of the BancGroup Common Stock, an appropriate and proportionate adjustment shall be made in the number of shares of BancGroup Common Stock into which the Acquired Corporation Stock shall be converted.
- 3.5 BancGroup Stock. The shares of Common Stock of BancGroup issued and outstanding immediately before the Effective Date shall continue to be issued and outstanding shares of the Resulting Corporation.
- 3.6 Dissenting Rights. Any shareholder of Acquired Corporation who shall have, prior to the vote with respect to approval of the Merger, furnished notice of intent to demand payment for his shares if the Merger is approved, who shall not have voted such shares in favor of this Agreement and who shall have complied with all other requirements set forth in the FBCA relating to the establishmen: of preservation of rights of dissenting shareholders, shall be entitled to receive payment for the fair value of his Acquired Corporation Stock. If after the Effective Date a dissenting shareholder of Acquired Corporation fails to perfect, or effectively withdraws or loses, his right to appraisal and payment for his shares of Acquired Corporation Stock, BancGroup shall issue and deliver the consideration to which such holder of shares of Acquired Corporation Stock is entitled under Section 3.1 (without interest) upon surrender of such holder of the certificate or certificates representing shares of Acquired Corporation Stock held by him.

A . . TICLE 4

REPRESENTATIONS, WARRANTIES AND COVENANTS OF BANCGROUP

BancGroup represents, warrants and covenants to and with Acquired Corporation as follows:

- 4.1 Organization. BancGroup is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. BancGroup has the necessary corporate powers to carry on its business as presently conducted and is qualified to do business in every jurisdiction in which the character and location of the Assets owned by it or the nature of the business transacted by it requires qualification or in which the failure to qualify could, individually or in the aggregate, have a Material Adverse Effect.
 - 4.2 Capital Stock.
- (a) The authorized capital stock of BancGroup consists of (A) 44,000,000 shares of Common Stock, \$2.50 par value per share, of which as of September 19, 1996, 16,296,558 shares were validly issued and outstanding, fully paid and nonassessable and are not subject to preemptive rights (not counting additional shares subject to issue pursuant to stock option and other plans and convertible debentures), and (B) 1,000,000 shares of Preference Stock, \$2.50 par value per share, none of which are issued and outstanding. The shares of BancGroup Common Stock to be issued in the Merger are duly authorized and, when so issued, will be validly issued and outstanding, fully paid and nonassessable, will have been registered under the 1933 Act, and will have been registered or qualified under the securities laws of all jurisdictions in which such registration or qualification is required, based upon information provided by Acquired Corporation.
- (b) The authorized capital stock of each Subsidiary of BancGroup is validly issued and outstanding, fully paid and nonassessable, and each Subsidiary is wholly owned, directly or indirectly, by BancGroup.
- 4.3 Financial Statements; Taxes. (a) BancGroup has delivered to Acquired Corporation copies of the following financial statements of BancGroup.
 - (i) Consolidated balance sheets as of December 31, 1994, and December 31, 1995, and for the nine months ended September 30, 1996;

- (ii) Consolidated statements of operations for each of the three years ended December 31, 1993, 1994 and 1995, and for the nine months ended September 30, 1996;
- (iii) Consolidated statements of cash flows for each of the three years ended December 31, 1993, 1994 and 1995, and for the nine months ended September 30, 1996; and
- (iv) Consolidated statements of changes in shareholders' equity for the three years ended December 31, 1993, 1994 and 1995, and for the nine months ended September 30, 1996.

All such financial statements are in all material respects in accordance with the books and records of BaneGroup and have been prepared in accordance with generally accepted accounting princip'es applied on a consistent basis throughout the periods indicated, all as more particularly set forth in the notes to such statements. Each of the consolidated balance sheets presents fairly as of its date the consolidated financial condition of BaneGroup and its Subsidiaries. Except as and to the extent reflected or reserved against in such balance sheets (including the notes thereto), BaneGroup did not have, as of the dates of such balance sheets, any material Liabilities or obligations (absolute or contingent) of a nature customarily reflected in a balance sheet or the notes thereto. The statements of consolidated income, shareholders' equity and changes in consolidated financial position present fairly the results of operations and changes in financial position of BaneGroup and its Subsidiaries for the periods indicated. The foregoing representations, insofar as they relate to the unaudited interim financial statements of BaneGroup for the nine months ended September 30, 1996, are subject in all cases to normal recurring year-end adjustments and the omission of footnote disclosure.

- (b) All Tax returns required to be filed by or on behalf of BaneGroup have been timely filed (or reques for extensions therefor have been timely filed and granted and have not expired), and all returns filed are complete and accurate in all material respects. All Taxes shown on these returns to be due and all additional assessments received have been paid. The amounts recorded for Taxes on the balance sheets provided under section 4.3(a) are, to the Knowledge of BancGroup, sufficient in all material respects for the payment of all unpaid federal, state, county, local, foreign or other Taxes (including any interest or penalties) of BancGroup accrued for or applicable to the period ended on the dates thereof, and all years and periods prior thereto and for which BancGroup may at such dates have been liable in its own right or as transferee of the Assets of, or as successor to, any other corporation or other party. No audit, examination or investigation is presently being inducted or, to the Knowledge of BancGroup, threatened by any taxing authority which is likely to result in a material Tax Liability, no material unpaid Tax deficiencies or additional liabilities of any sort have been proposed by any governmental representative and no agreements for extension of time for the assessment of any material amount of Tax have been entered into by or on behalf of BancGroup. BancGroup has withheld from its employees (and timely paid to the appropriate governmental entity) proper and accurate amounts for all periods in material compliance with all Tax withholding provisions of applicable federal, state, foreign and local Laws (including without limitation, income, social security and employment Tax withholding for all types of compensation).
- 4.4 No Conflict with Other Instrument. The consummation of the transactions contemplated by this Agreement will not result in a breach of or constitute a Default (without regard to the giving of notice or the passage of time) under any material Contract, indenture, mortgage, deed of trust or other material agreement or instrument to which BancGroup or any of its Subsidiaries is a party or by which they or their Assets may be bound; will not conflict with any provision of the restated certificate of incorporation or bylaws of BancGroup or the articles of incorporation or bylaws of any of its Subsidiaries; and will not violate any provision of any Law, regulation, judgment or decree binding on them or any of their Assets.
- 4.5 Absence of Material Adverse Change. Since the date of the most recent balance sheet provided under section 4.3(a)(i) above, there have been no events, changes or occurrences which have had or are reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on BancGroup.
- 4.6 Approval of Agreements. The board of directors of BancGroup has approved this Agreement and the transactions contemplated by it and has authorized the execution and delivery by BancGroup of this Agreement. This Agreement constitutes the legal, valid and binding obligation of BancGroup, enforceable against it in accordance with its terms. Approval of this Agreement by the stockholders of BancGroup is not

required by applicable law. Subject to the matters referred to in section 8.2, BancGroup has full power, authority and legal right to enter into this Agreement and to consummate the transactions contemplated by this Agreement. BancGroup has no Knowledge of any fact or circumstance under which the appropriate regulatory approvals required by section 8.2 will not be granted without the imposition of material conditions or material delays.

- 4.7 Tax Treatment. BancGroup has no present plan to sell or otherwise dispose of any of the Assets of Acquired Corporation, subsequent to the Merger, and BancGroup intends to continue the historic business of Acquired Corporation.
- 4.8 Title and Related Matters. BancGroup has good and marketable title to all the properties, interests in properties and Assets, real and personal, reflected in the most recent balance sheet referred to in section 4.3(a), or acquired after the date of such balance sheet (except properties, interests and Assets sold or otherwise disposed of since such date, in the ordinary course of business), free and clear of all mortgages, Liens, pledges, charges or encumbrances except (i) mortgages and other encumbrances referred to in the notes of such balance sheet, (ii) liens for current Taxes not yet due and payable and (iii) such imperfections of title and easements as do not materially detract from or interfere with the present use of the properties subject thereto or affected thereby, or otherwise materially impair present business operations at such properties. To the Knowledge of BancGroup, the material structures and equipment of BancGroup comply in all material respects with the requirements of all applicable Laws.
- 4.9 Sub: idiaries. Each Subsidiary of BancGroup has been duly incorporated and is validly existing as a corporation in good standing under the Laws of the jurisdiction of its incorporation and each Subsidiary has been duly qualified as a foreign corporation to transact business and is in good standing under the Laws of each other jurisdiction in which it owns or leases properties, or conducts any business so as to require such qualification and in which the failure to be duly qualified could have a Material Adverse Effect upon BancGroup and its Subsidiaries considered as one enterprise; each of the banking Subsidiaries of BancGroup has its deposits fully insured by the Federal Deposit Insurance Corporation to the extent provided by the Federal Deposit Insurance Act; and the businesses of the non-bank Subsidiaries of BancGroup are permitted to subsidiaries of registered bank holding companies.
- 4.10 Contracts. Neither BancGroup nor any of its Subsidiaries is in violation of its respective certificate of incorporation or bylaws or in Default in the performance or observance of any material obligation, agreement, covenant or condition contained in any Contract, indenture, mortgage, loan agreement, note, lease or other instrument to which it is a party or by which it or its property may be bound.
- 4.11 Litigation. Except as disclosed in or reserved for in BancGroup's financial statements, there is no Litigation before or by any court or Agency, domestic or foreign, now pending, or, to the Knowledge of BancGroup, threatened against or affecting BancGroup or any of its Subsidiaries (nor is BancGroup aware of any facts which could give rise to any such Litigation) which is required to be disclosed in the Registration Statement (other than as disclosed therein), or which is likely to have any Material Adverse Effect or prospective Material Adverse Effect, or which is likely to materially and adversely affect the properties or Assets thereof or which is likely to materially affect or delay the consummation of the transactions contemplated by this Agreement; all pending legal or governmental proceedings to which BancGroup or any Subsidiary is a party or of which any of their properties is the subject which are not described in the Registration Statement, including ordinary routine litigation incidental to the business, are, considered in the aggregate, not material; and neither BancGroup and its Subsidiaries have any contingent obligations which could be considered material to BancGroup and its Subsidiaries considered as one enterprise which are not disclosed in the Registration Statement as it may be amended or supplemented.
- 4.12 Compliance. BancGroup and its Subsidiaries, in the conduct of their businesses, are to the Knowledge of BancGroup, in material compliance with all material federal, state or local Laws applicable to their or the conduct of their businesses.
- 4.13 Registration Statement. At the time the Registration Statement becomes effective and at the time of the Stockholders' Meeting, the Registration Statement, including the Proxy Statement which shall

constitute a part thereof, will complete in all material respects with the requirements of the 1933 Act and the rules and regulations thereunder, but contain an untrue statement of a material fact or omit to state a material fact necessary in order to a like statements therein, in the light of the circumstances under which they were made, not misleading; prov. It is however, that the representations and warranties in this subsection shall not apply to statements in or omissions from the Proxy Statement made in reliance upon and in conformity with information furnished in writing to BaneGroup by Acquired Corporation or any of its representatives expressly for use in the Proxy Statement or information included in the Proxy Statement regarding the business of Acquired Corporation, its operations, Assets and capital.

- 4.14 SEC Fillings. (a) BancGroup has heretofore delivered to Acquired Corporation copies of Banc-Group's: (i) Annual Report on Form 10-K for the fiscal year ended December 31, 1995; (ii) 1995 Annual Report to Shareholders; (iii) Quarterly Report on Form 10-Q for the fiscal quarters ended March 31, June 30, and September 30, 1996; and (iv) any reports on Form 8-K, filed by BancGroup with the SEC since December 31, 1995. Since December 31, 1995, BancGroup has timely filed all reports and registration statements and the documents required to be filed with the SEC under the rules and regulations of the SEC and all such reports and registration statements or other documents have complied in all material respects, as of their respective filing dates and effective dates, as the case may be, with all the applicable requirements of the 1933 Act and the 1934 Act. As of the respective filing and effective dates, none of such reports or registration statements or other documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (b) The documents incorporated by reference into the Registration Statement, at the time they were filed with the SEC, complied in all material respects with the requirements of the 1934 Act and Regulations thereunder and when read together and with the other information in the Registration Statement will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading at the time the Registration Statement becomes effective or at the time of the Stockholder Meeting.
- 4.15 Form S-4. The conditions for use of a registration statement on SEC Form S-4 set forth in the General instructions on Form S-4 have been or will be satisfied with respect to BancGroup and the Registration Statement.
- 4.16 Brokers. All negotiations relative to this Agreement and the transactions contemplated by this Agreement have been carried on by BancGroup directly with Acquired Corporation and without the intervention of any other person, either as a result of any act of BancGroup or otherwise in such manner as to give rights to any valid claim against BancGroup for finders fees, brokerage commissions or other like payments.
- 4.17 Government Authorization. BancGroup and its Subsidiaries have all Permits that, to the Knowledge of BancGroup and its Subsidiaries, are or will be legally required to enable BancGroup or any of its Subsidiaries to conduct their businesses in all material respects as now conducted by each of them.
- 4.18 Absence of Regulatory Communications. Neither BancGroup nor any of its Subsidiaries is subject to, or has received during the past three (3) years, any written communication directed specifically to it from any Agency to which it is subject, pursuant to which such Agency has imposed or has indicated it may impose any material restrictions on the operations of, or the business conducted by, such company or in which such Agency has raised a material question concerning the condition, financial or otherwise, of such company.
- 4.19 Disclosure. No representation or warranty, or any statement or certificate furnished or to be furnished to Acquired Corporation by BancGroup, contains or will contain any untrue statement of a material fact, or omits or will omit to state a material fact necessary to make the statements contained in this Agreement or in any such statement or certificate not misleading.

ARTICLE 5

REPRESENTATIONS, WARRANTIES AND COVENANTS OF ACQUIRED CORPORATION

Acquired Corporation represents, warrants and covenants to and with BancGroup, as follows:

- 5.1 Organization. Acquired Corporation is a Florida corporation, and the Bank is a Florida state bank. Each Acquired Corporation Company is duly organized, validly existing and in good standing under the respective Laws of its jurisdiction of incorporation and has all requisite power and authority to carry on its business as it is now being conducted and is qualified to do business in every jurisdiction in which the character and location of the Assets owned by it or the nature of the business transacted by it requires qualification or in which the failure to qualify could, individually, or in the aggregate, have a Material Adverse Effect.
- 5.2 Capital Stock. (i) As of October 31, 1996, the authorized capital stock of Acquired Corporation consisted of 10,000,000 shares of common stock, \$.001 par value per share, 990,553 shares of which are issued and outstanding. All of such shares which are outstanding are validly issued, fully paid and nonassessable and not subject to preemptive rights. Acquired Corporation has 60,410 shares of its common stock subject to cteroise at any time pursuant to Acquired Corporation Options. Except for the foregoing, Acquired Corporation does not have any other arrangements or commitments obligating it to issue shares of its capital stock or any securities convertible into or having the right to purchase shares of its capital stock, including the granting of additional options to purchase its common stock.
- 5.3 Subsidiaries. Acquired Corporation has no direct Subsidiaries other than the Bank, and there are no Subsidiaries of the Bank. Acquired Corporation owns all of the issued and outstanding capital stock of the Bank free and clear of any liens, claims or encumbrances of any kind. All of the issued and outstanding shares of capital stock of the Subsidiaries have been validly issued and are fully paid and non-assessable. As of October 31, 1596, there were 2,000,000 shares of the common stock, par value \$8.00 per shared uthorized of the Bank, 990,553 of which are issued and outstanding and wholly owned by Acquired Corporation. The Bank has no arrangements or commitments obligating it to issue shares of its capital stock or any securities convertible into or having the right to purchase shares of its capital stock.
- 5.4 Financial Statements; Taxes (a) Acquired Corporation has delivered to BancGroup copies of the following financial statements of Acquired Corporation:
 - (i) Consolidated statements of financial condition as of December 31, 1995 and 1996, and for the nine months ended September 30, 1996;
 - (ii) Consolidated statements of income for each of the three years ended December 31, 1994, 1995 and 1996, and for the nine months ended September 30, 1996;
 - (iii) Consolidated statements of stockholders' equity for each of the three years ended December 31, 1994, 1995, and 1996, and for the nine months ended September 30, 1996; and
 - (iv) Consolidated statements of cash flows for the three years ended December 31, 1994, 1995 and 1996, and for the nine months ended September 30, 1996.

All of the foregoing financial statements are in all material respects in accordance with the books and records of Acquired Corporation and have been prepared in accordance with generally accepted accounting principles applied on a consistent basis throughout the periods indicated, except for changes required by GAAP, all as more particularly set forth in the notes to such statements. Each of such balance sheets presents fairly as of its date the financial condition of Acquired Corporation. Except as and to the extent reflected or reserved against in such balance sheets (including the notes thereto), Acquired Corporation did not have, as of the date of such balance sheets, any material Liabilities or obligations (absolute or contingent) of a nature customarily reflected in a balance sheet or the notes thereto. The statements of income, stockholders' equity and cash flows present fairly the results of operation, changes in shareholders equity and cash flows of Acquired Corporation for the periods indicated. The foregoing representations, insofar as they relate to the unaudited interim financial statements of Acquired Corporation for the nine months ended September 30,

1996, are subject in all cases to normal recurring year-end adjustments and the omission of footnote disclosure.

- (b) Except as set forth on Schedule 5.4(b), all Tax returns required to be filed by or on behalf of Acquired Corporation have been timely filed (or requests for extensions therefor have been timely filed and granted and have not expired), and all returns filed are complete and accurate in all material respects. All Taxes shown on these returns to be due and all additional assessments received have been paid. The amounts recorded for Taxes on the balance sheets provided under section 5.4(a) are, to the Knowledge of Acquired Corporation, sufficient in all material respects for the payment of all unpaid federal, state, county, local, foreign and other Taxes (including any interest or penalties) of Acquired Corporation accrued for or applicable to the period ended on the dates thereof, and all years and periods prior the etc and for which Acquired Corporation may at such dates have been liable in its own right or at a transferce of the Assets of, or as successor to, any other corporation or other party. No audit, examination or investigation is presently being conducted or, to the Knowledge of Acquired Corporation, threatened by any taxing authority which is likely to result in a material Tax Liability, no material unpaid Tax deficiencies or additional liability of any sort have been proposed by any governmental representative and no agreements for extension of time for the assessment of any material amount of Tax have been entered into by or on behalf of Acquired Corporation. Acquired Corporation has not executed an extension or waiver of any statute of limitations on the assessment or collection of any Tax due that is currently in " rot.
- (c) Each Acquired Corporation Company has withheld from its employees (and timely paid to the appropriate governmental entity) proper and accurate amounts for all periods in material compliance with all Tax withholding provisions of applicable federal, state, foreign and local Laws (including without limitation, income, social security and employment Tax withholding for all types of compensation). Each Acquired Corporation Company is in compliance with, and its records contain all information and documents (including properly completed IRS Forms W-9) necessary to comply with, all applicable information reporting and Tax withholding requirements under federal, state and local Tax Laws, and such records identify with specificity all accounts subject to backup withholding under section 3406 of the Code.
- 5.5 Absence of Certain Changes or Events. Except as set forth on Schedule 5.5, since the date of the most recent balance sheet provided under section 5.4(a) (i) above, no Acquired Corporation Company has
 - (a) issued, delivered or agreed to issue or deliver any stock, bonds or other corporate securities (whether authorized and unissued or held in the treasury) except shares of common stock issued or to be issued upon the exercise of Acquired Corporation Options existing on the date hereof;
 - (b) borrowed or agreed to borrow any funds or incurred, or become subject to, any Liability (absolute or contingent) except borrowings, obligations (including purchase of federal funds) and Liabilities incurred in the ordinary course of business and consistent with past practice;
 - (c) paid any material obligation or Liability (absolute or contingent) other than current Liabilities reflected in or shown on the most recent balance sheet referred to in section 5.4(a) (i) and current Liabilities incurred since that date in the ordinary course of business and consistent with past practice;
 - (d) declared or made, or agreed to declare or make, any payment of dividends or distributions of any Assets of any kind whatsoever to shareholders, or purchased or redeemed, or agreed to purchase or redeem, any of its outstanding securities, provided, however, that Acquired Corporation shall be permitted to pay a cash dividend to its shareholders for the fourth quarter of 1996 and the first quarter of 1997. Such dividend shall be equal to \$Divicend X .88732, where \$Dividend equals the per share dividend paid by BancGroup for that quarter in question;
 - (e) except in the ordinary course of business, sold or transferred, or agreed to sell or transfer, any of its Assets, or canceled, or agreed to cancel, any debts or claims;
 - (f) except in the ordinary course of business, entered or agreed to enter into any agreement or arrangement granting any preferential rights to purchase any of its Assets, or requiring the consent of any party to the transfer and assignment of any of its Assets;

- (g) suffered any Losses or waived any rights of value which in either event in the aggregate are material considering its business as u whole;
- (h) except in the ordinary course of business, made or permitted any amendment or termination of any Contract, agreement or license to which it is a party if such amendment or termination is material considering its business as a whole;
- (i) except in accordance with normal and usual practice, made any accrual or arrangement for or payment of honuses or special compensation of any kind or any severance or termination pay to any present or former officer or employee;
- (j) except in accordance with normal and usual practice, increased the rate of compensation payable to or to become payable to any of its officers or employees or made any material increase in any profit sharing, bonus, deferred compensation, savings, insurance, pension, retirement or other employee benefit plan, payment or arrangement made to, for or with any of its officers or employees;
- (k) received notice or had Knowledge or reason to believe that any of its substantial customers has terminated or intends to terminate its relationship, which termination would have a Material Adverse Effect on its financial condition, results of operations, business, Assets or properties;
- (1) failed to operate its business in the ordinary course so as to preserve its business intact and to preserve the goodwill of its customers and others with whom it has business relations;
 - (m) entered into any other material transaction other than in the ordinary course of business; or
 - (n) agreed in writing, or otherwise, to take any action described in clauses (a) through (m) above.

Between the date hereof and the Effective Date, no Acquired Corporation Company, without the express written approval of BancGroup, will do any of the things listed in clauses (a) through (n) of this section 5.5 except as permitted therein or as contemplated in this Agreement, and no Acquired Corporation Company will enter into or amend any material Contract, other than Loans or renewals thereof entered into in the ordinary course of business, without the express written consent of BancGroup.

5.6 Title and Related Matters.

- (a) Title. Acquired Corporation has good and marketable title to all the properties, interest in properties and Assets, real and personal, reflected in the most recent balance sheet referred to in section 5.4(a) (i), or acquired after the date of such balance sheet (except properties, interests and Assets sold or otherwise disposed of since such date, in the ordinary course of business), free and clear of all mortgages, Liens, pledges, charges or encumbrances except (i) mortgages and other encumbrances referred to in the notes to such balance sheet, (ii) Liens for current Taxes not yet due and payable and (iii) such imperfections of title and easements as do not materially detract from or interfere with the present use of the properties subject thereto or affected thereby, or otherwise materially impair present business operations at such properties. The Acquired Corporation has no Knowledge that any of the material structures and equipment of each Acquired Corporation Company fail to comply in any material respect with the requirements of all applicable Laws.
- (b) Leases. Schedule 5.6(b) sets forth a list and description of all real and personal property owned or leased by any Acquired Corporation Company, either as lessor or lessee.
- (c) Personal Property. Schedule 5.6(c) sets forth a depreciation schedule of each Acquired Corporation Company's fixed Assets as of October 31, 1996.
- (d) Computer Hardware and Software. Schedule 5.6(d) contains a description of all agreements relating to data processing computer software and hardware now being used in the business operations of any Acquired Corporation Company. Acquired Corporation is not aware of any defects, irregularities or problems with any of its computer hardware or software which renders such hardware or software unable to satisfactorily perform the tasks and functions to be performed by them in the business of any Acquired Corporation Company.

- 5.7 Commitments. Except as set forth in Schedule 5.7, no Acquired Corporation Company is a party to any oral or written (i) Contracts for the employment of any officer or employee which is not terminable on 30 days' (or less) notice, (ii) profit sharing, bonus, deferred compensation, savings, stock option, severance pay, pension or retirement plan, agreement or arrangement, (iii) loan agreement, indenture or similar agreement relating to the horrowing of money by such party, (iv) guaranty of any obligation for the borrowing of money or otherwise, excluding endorsements made for collection, and guaranties made in the ordinary course of business, (v) consulting or other similar material Contracts, (vi) collective bargaining agreement, (vii) agreement with any present or former officer, director or shareholder of such party, or (viii) other Contract, agreement or other commitment which is material to the business, operations, property, prospects or Assets or to the condition, financial or otherwise, of any Acquired Corporation Company, Complete and accurate copies of all Contracts, plans and other items so listed have been made or will be made available to BancGroup for inspection.
- 5.8 Charter and Bylaws. Schedule 5.8 contains true and correct copies of the articles of incorporation and bylaws of each Acquired Corporation Company, including all amendments thereto, as currently in effect. There will be no changes in such articles of incorporation or bylaws prior to the Effective Date, without the prior written consent of BancGroup.
- 5.9 Litigation. There is no Litigation (whether or not purportedly on behalf of Acquired Corporation) pending or, to the Knowledge of Acquired Corporation, threatened against or affecting any Acquired Corporation Company (nor is Acquired Corporation aware of any facts which are likely to give rise to any such Litigation) at law or in equity, or before or by any governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, or before any arbitrator of any kind, which involves the possibility of any judgment or Liability not fully covered by insurance in excess of a reasonable deductible amount or which may have a Material Adverse Effect on Acquired Corporation, and no Acquired Corporation Company is in Default with respect to any judgment, order, writ, injunction, decree, award, rule or regulation of any court, arbitrator or governmental department, commission, board, bureau, agency or instrumentality, which Default would have a Material Adverse Effect on Acquired Corporation. To the Knowledge of Acquired Corporation, each Acquired Corporation Company has complied in all material respects with all material applicable Laws and Regulations including those imposing Taxes, of any applicable jurisdiction and of all states, municipalities, other political subdivicums and Agencies, in respect of the ownership of its properties and the conduct of its business, which, if not complied with, would have a Material Adverse Effect on Acquired Corporation.
- 5.10 Material Contract Defaults. Except as disclosed on Schedule 5.10, no Acquired Corporation Company is in Default in any material respect under the terms of any material Contract, agreement, lease or other commitment which is or may be material to the business, operations, properties or Assets, or the condition, financial or otherwise, of such company and the Acquired Corporation has no Knowledge of any event which, with notice or lapse of time, or both, may be or become an event of Default under any such material Contract, agreement, lease or other commitment in respect of which adequate steps have not been taken to prevent such a Default from occurring.
- 5.11 No Conflict with Other Instrument. The consummation of the transactions contemplated by this Agreement will not result in the breach of any term or provision of or constitute a Default under any material Contract, indenture, mortgage, deed of trust or other material agreement or instrument to which any Acquired Corporation Company is a party and will not conflict with any provision of the charter or bylaws of any Acquired Corporation Company.
- 5.12 Governmental Authorization. Each Acquired Corporation Company has all Permits that, to the Knowledge of Acquired Corporation, are or will be legally required to enable any Acquired Corporation Company to conduct its business in all material respects as now conducted by each Acquired Corporation Company.
- 5.13 Absence of Regulatory Communications. Except as provided in Schedule 5.13, no Acquired Corporation Company is subject to, nor has any Acquired Corporation Company received during the past three years, any written communication directed specifically to it from any Agency to which it is subject,

pursuant to which such Agency has imposed or has indicated it may impose any material restrictions on the operations of, or the business conducted by, such company or in which such Agency has raised any material question concerning the condition, financial or otherwise, of such company.

- 5.14 Absence of Material Adverse Change. To the Knowledge of Acquired Corporation, since the date of the most recent balance sheet provided under section 5.4(a) (i), there have been no events, changes or occurrences which have had, or are reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on any Acquired Corporation Company.
- 5.15 Insurance. Each Acquired Corporation Company has in effect insurance coverage and bonds with reputable insurers which, in respect to amounts, types and risks insured, management of Acquired Corporation reasonably believes to be adequate for the type of business conducted by such company. No Acquired Corporation Company is liable for any material retroactive premium adjustment. All insurance policies and bonds are valid, enforceable and in full force and effect, and no Acquired Corporation Company has received any notice of any material premium increase or cancellation with respect to any of its insurance policies or bonds. Within the last three years, no Acquired Corporation Company has been refused any insurance coverage which it has sought or applied for, and it has no reason to believe that existing insurance coverage cannot be renewed as and when the same shall expire, upon terms and conditions as favorable as those presently in effect, other than possible increases in premiums that do not result from any extraordinary loss experience. All policies of insurance presently held or policies containing substantially equivalent coverage will be outstanding and in full force with respect to each Acquired Corporation Company at all times from the date hereof to the Effective Date.

5.16 Pension and Employee Benefit Plans.

- (a) To the Knowledge of Acquired Corporation, all employee benefit plans of each Acquired Corporation Company have been established in compliance with, and such plans have been operated in material compliance with, all applicable Laws. Except as set forth in Schedule 5.16, no Acquired Corporation Company sponsors or otherwise maintains a "pension plan" within the meaning of section 3(2) of ERISA or any other retirement plan other than the Bank's 401k Profit Sharing Plan that is intended to qualify under section 401 of the Code, nor do any unfunded Liabilities exist with respect to any employee benefit plan, past or present. To the Knowledge of Acquired Corporation, no employee benefit plan, any trust created thereunder or any trustee or administrator thereof has engaged in a "prohibited transaction," as defined in section 4975 of the Code, which may have a Material Adverse Effect on the condition, financial or otherwise, of any Acquired Corporation Company.
- (b) To the Knowledge of Acquired Corporation, no amounts payable to any employee of any Acquired Corporation Company will fail to be deductible for federal income tax purposes by virtue of Section 280G of the Code and regulations thereunder.
- 5.17 Buy-Sell Agreement. To the Knowledge of Acquired Corporation, there are no agreements among any of its shareholders granting to any person or persons a right of first refusal in respect of the sale, transfer, or other disposition of shares of outstanding securities by any shareholder of Acquired Corporation, any similar agreement or any voting agreement or voting trust in respect of any such shares.
- 5.18 Brokers. All negotiations relative to this Agreement and the transactions contemplated by this segment have been carried on by Acquired Corporation directly with BancGroup and without the ention of any other person, either as a result of any act of Acquired Corporation, or otherwise, in such near as to give rise to any valid claim against Acquired Corporation for a finder's fee, brokerage commission or other like payment.
- 5.19 Approval of Agreements. The board of directors of Acquired Corporation has approved this Agreement and the transactions contemplated by this Agreement and has authorized the execution and delivery by Acquired Corporation of this Agreement. Subject to the matters referred to in section 8.2, Acquired Corporation has full power, authority and legal right to enter into this Agreement, and, upon appropriate vote of the shareholders of Acquired Corporation in accordance with this Agreement, Acquired

Corporation shall have full power, authority and legal right to consummate the transactions contemplated by this Agreement.

- 5.20 Disclosure. No representation or warranty, nor any statement or certificate furnished or to be furnished to BancGroup by Acquired Corporation, contains or will contain any untrue statement of a material fact, or omits or will omit to state a material fact necessary to make the statements contained in this Agreement or in any such statement or certificate not misleading.
- 5.21 Registration Statement. At the time the Registration Statement becomes effective and at the time of the Stockholders Meeting, the Registration Statement, including the Proxy Statement which shall constitute part thereof, will not contain an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading; provided, however, that the representations and warranties in this section shall only apply to statements in or omissions from the Proxy Statement relating to descriptions of the business of Acquired Corporation, its Assets, properties, operations, and capital stock or to information furnished in writing by Acquired Corporation or its representatives expressly for inclusion in the Proxy Statement.
- 5.22 Loans; Adequacy of Allowance for Loan Losses. All reserves for loan losses shown on the most recent financial statements furnished by Acquired Corporation have been calculated in accordance with prudent and customary banking practices and are adequate in all material respects to reflect the risk inherent in the loans of any Acquired Corporation Company. Acquired Corporation has no Knowledge of any fact which is likely to require a future material increase in the provision for 1 an losses or a material decrease in the loan loss reserve reflected in such financial statements. Each loan reflected as an Asset on the financial statements of Acquired Corporation is the legal, valid and binding obligation of the obligor of each loan, enforceable in accordance with its terms subject to the effect of bankruptcy, insolvency, reorganization, moratorium, or other similar laws relating to creditors' rights generally and to general equitable principles. No Acquired Corporation Company has in its portfolio any loan exceeding its legal lending limit, nor, except as disclosed on Schedule 5.22, any known significant delinquent, substandard, doubtful, loss, nonperforming or problem loans.
- 5.23 Environmental Matters. Except as provided in Schedule 5.23, to the Knowledge of Acquired Corporation, each Acquired Corporation Company is in material compliance with all Laws and other governmental requirements relating to the generation, management, handling, transportation, treatment, disposal, storage, delivery, discharge, release or emission of any waste, pollution, or toxic, hazardous or other substance (the "Environmental Laws"), and Acquired Corporation has no Knowledge that any Acquired Corporation Company has not complied with all regulations and requirements promulgated by the Occupational Safety and Health Administration that are applicable to any Acquired Corporation Company. To the Knowledge of Acquired Corporation, there is no Litigation pending or threatened with respect to any violation or alleged violation of the Environmental Laws. Acquired Corporation has no Knowledge, with respect to Assets of or owned by any Acquired Corporation Company, that (i) there has been spillage, leakage, contamination or release of any substances for which the appropriate remedial action has not been completed; (ii) any owned or leased property is contaminated with or contains any hazardous substance or waste; and (iii) there are any underground storage tanks on any premises owned or leased by any Acquired Corporation Company. Acquired Corporation has no Knowledge of any facts which might suggest that any Acquired Corporation Company has engaged in any management practice with respect to any of its past or existing borrowers which could reasonably be expected to subject any Acquired Corporation Company to any Liability, either directly or indirectly, under the principles of law as set forth in United States v. Fleet Factors Corp., 901 F.2d 1550 (11th Cir. 1990) or any similar principles. Moreover, Acquired Corporation has no Knowledge that any Acquired Corporation Company has extended credit, either on a secured or unsecured basis, to any person or other entity engaged in any activities which would require or requires such person or entity to obtain any Permit under any Environmental Law which has not been obtained.
- 5.24 Transfer of Shares. Acquired Corporation has no Knowledge of any plan or intention on the part of Acquired Corporation's shareholders to sell or otherwise dispose of any of the BancGroup Common Stock to be received by them in the Merger that would reduce such shareholders' ownership to a number of shares

having, in the aggregate, a fair market value of less than fifty (50%) percent of the total fair market value of Acquired Corporation common stock outstanding immediately before the Merger.

- 5.25 Collective Bargaining. There are no labor contracts, collective bargaining agreements, letters of undertakings or other arrangements, formal or informal, between any Acquired Corporation Company and any union or labor organization covering any of Acquired Corporation Company's employees and some of said employees are represented by any union or labor organization.
- 5.26 Labor Disputes. To the Knowledge of Acquired Corporation, each Acquired Corporation Company is in material compliance with all federal and state laws respecting employment and employment practices, terms and conditions of employment, wages and hours. No Acquired Corporation Company is or has been engaged in any unfair labor practice, and, to the Knowledge of Acquired Corporation, no unfair labor practice complaint against any Acquired Corporation Company is pending before the National Labor Relations Board. Relations between management of each Acquired Corporation Company and the employees are amicable and there have not been, nor to the Knowledge of Acquired Corporation, are there presently, any attempts to organize employees, nor to the Knowledge of Acquired Corporation, are there plans for any such attempts.
- 5.27 Derivative Contracts. No Acquired Corporation Company is a party to or has agreed to enter into a swap, forward, future, option, cap, flour or collar financial contract, or any other interest rate or foreign currency protection contract or derivative security not included in Acquired Corporation's financial statements delivered under section 5.4 hereof which is a financial derivative contract (including various combinations thereof).

ARTICLE 6

ADDITIONAL COVENANTS

- 6.1 Additional Covenants of BancGroup. BancGroup covenants to and with Acquired Corporation as follows:
 - (a) Registration Statement and Other Filings. BancGroup shall prepare and file with the SEC the Registration Statement on Form S-4 (or such other form as may be appropriate) and all amendments and supplements thereto, in form reasonably satisfactory to Acquired Corporation and its counsel, with respect to the Common Stock to be issued pursuant to this Agreement. BancGroup shall use reasonable good faith efforts to prepare all necessary filings with any Agencies which may be necessary for approval to consummate the transactions contemplated by this Agreement.
 - (b) Blue Sky Permits. BancGroup shall use its best efforts to obtain, prior to the effective date of the Registration Statement, all necessary state securities Law or "blue sky" Permits and approvals required to carry out the transactions contemplated by this Agreement.
 - (c) Financial Statements. BancGroup shall furnish to Acquired Corporation:
 - (i) As soon as practicable and in any event within forty-five (45) days after the end of each quarterly period (other than the last quarterly period) in each fiscal year, consolidated statements of operations of BancGroup for such period and for the period beginning at the commencement of the fiscal year and ending at the end of such quarterly period, and a consolidated statement of financial condition of BancGroup as of the end of such quarterly period, setting forth in each case in comparative form figures for the corresponding periods ending in the preceding fiscal year, subject to changes resulting from year-end adjustments;
 - (ii) Promptly upon receipt thereof, copies of all audit reports submitted to BancGroup by independent auditors in connection with each annual, interim or special audit of the books of BancGroup made by such accountants;

- (iii) As soon as practicable, copies of all such financial statements and reports as it shall send to its stockholders and of such regular and periodic reports as BaneGroup may file with the SEC or any other Agency; and
- (iv) With reasonable promptness, such additional financial data as Acquired Corporation may reasonably request.
- (d) No Control of Acquired Corporation by BancGroup. Notwithstanding any other provision hereof, until the Effective Date, the authority to establish and implement the business policies of Acquired Corporation shall continue to reside solely in Acquired Corporation's officers and board of directors.
- (a) Listing. Prior to the Effective Date, BancGroup shall use its reasonable efforts to list the shares of BancGroup Common Stock to be issued in the Merger on the NYSE or other quotations system on which such shares are primarily traded.
- (f) Employee Benefit Matters. On the Effective Date, all employees of any Acquired Corporation Company shall, at BancGroup's option, either become employees of the Resulting Corporation or its Subsidiaries or be entitled to severance benefits in accordance with Colonial Bank's severance policy as of the date of this Agreement. All employees of any Acquired Corporation Company who become employees of the Resulting Corporation or its Subsidiaries on the Effective Date shall be entitled, to the extent permitted by applicable Law, to participate in all benefit plans of Colonial Bank to the same extent as Colonial Bank employees, except as stated otherwise in this section. Employees of any Acquired Corporation Company who become employees of the Resulting Corporation or its Subsidiaries on the Effective Date shall be allowed to participate as of the Effective Date in the medical and dental benefits plan of Colonial Bank as new employees of Colonial Bank, and the time of employment of such employees who are employed at least 30 hours per week with any Acquired Corporation Company as of the Effective Date shall be counted as employment under such dental and medical plans of Colonial Bank for purposes of calculating any 30 day waiting period and pre-existing condition limitations. To the extent permitted by applicable Law, the period of service with the appropriate Acquired Corporation Company of all employees who become employees of the Resulting Corporation or its Subsidiaries on the Effective Date shall be recognized only for vesting and eligibility purposes under Colonial Bank's benefit plans. In addition, if the Effective Date falls within an annual period of coverage under any group health plan or group dental plan of the Resulting Corporation and its Subsidiaries, each such Acquired Corporation Company employee shall be given credit for covered expenses paid by that employee under comparable employee benefit plans of the Acquired Corporation Company during the applicable coverage period through the Effective Date towards satisfaction of any annual deductible limitation and out-of-pocket maximum that may apply under that group health plan or group dental plan of the Resulting Corporation and its Subsidiaries.
- (g) Acquisition of Discovery Period Insurance Coverage. BancGroup shall, or shall cause Colonial Bank to, use its reasonable best efforts to purchase a runoff "discovery period" extension under the Acquired Corporation's existing director and officer liability insurance policy covering persons who are currently covered by such insurance for a period of two years after the Effective Date; provided, that neither BancGroup nor Colonial Bank shall be obligated to pay more than \$20,000 in premium costs for such coverage.
- . (h) Indemnification. (i) Subject to the conditions set forth in the succeeding paragraph, for a period of five years after the Effective Date BaneGroup shall, and shall cause Colonial Bank to, indemnify, defend and hold harmless each person entitled to indemnification from the Acquired Corporation (each being an "Indemnified Party") against all liabilities arising out of actions or omissions occurring upon or prior to the Effective Date (including without limitation the transactions contemplated by this Agreement) to the maximum extent authorized under Section 607.0850 of the FBCA.
- (ii) Any Indemnified Party wishing to claim indemnification under this subsection (h), upon learning of any such liability or Litigation, shall promptly notify BancGroup thereof. In the event of any such Litigation

(whether arising before or after the Effective Date (i) BancGroup or Colonial Bank shall have the right to assume the defense thereof with counsel reasonably acceptable to such Indemnified Party and, upon assumption of such defense, BancGroup shall not be liable to such Indemnified Parties for any legal expenses of other counsel or any other expenses subsequently incurred by such Indemnified Parties in connection with the defense thereof, except that if BancGroup or Colonial Bank elects not to assume such defense or counsel for the Indemnified Parties advises that there are substantive issues which raise conflicts of interest between BancGroup and the Indemnified Parties, the Indemnified Parties may retain counsel satisfactory to them, and BancGroup or Colonial Bank shall pay all reasonable fees and expenses of such counsel for the Indemnified Parties promptly as statements therefor are received; provided that BancGroup shall be obligated pursuant to this subsection to pay for only one firm of counsel for all Indemnified Parties in any jurisdiction, (ii) the Indemnified Parties will cooperate in the defense of any such Litigation; and (iii) BancGroup shall not be liable for any settlement effected without its prior consent; and further provided that BancGroup and Colonial Bank shall not have any obligation hereunder to any Indemnified Party when and if a court of competent jurisdiction shall determine, and such determination shall have become final, that the indemnification of such Indemnified Party in the manner contemplated hereby is prohibited by applicable law.

- (iii) In consideration of and as a condition precedent to the effectiveness of the indemnification obligations provided by BancGroup in this section to a director or officer of the Acquired Corporation, such director or officer of the Acquired Corporation shall have delivered to BancGroup on or prior to the Effective Date a letter in form reasonably satisfactory to BancGroup concerning claims such directors or officer may have against Acquired Corporation. In the letter, the directors or officer shall: (i) acknowledge the assumption by BancGroup as of the Effective Date of all Liability (to the extent Acquired Corporation is so liable) for claims for indemnification arising under section 6.1(h) hereof; (ii) affirm that they do not have nor are they aware of any claims they might have (other than those referred to in the following clause (iii)) against Acquired Corporation; (iii) identify any claims or any facts or circumstances of which they are aware that could give rise to a claim for indemnification under section 6.1(h)(i) hereof; and (iv) release as of the Effective Date any and all claims that they may have against any Acquired Corporation Company other than (A) those referred to in the foregoing clause(iii) and disclosed in the letter of the director or officer, (B) claims which have not yet been asserted against such director or officer (other than claims arising from facts and circumstances of which such director or officer is aware but which are not disclosed in such director or executive officer's letter), (C) claims arising from any transaction contemplated by this Agreement or disclosed in any schedule to this Agreement, and (D) claims arising in the ordinary course of business of any Acquired Corporation Company after the date of the letter.
- (iv) Acquired Corporation hereby represents and warrants to BancGroup that it has no Knowledge of any claim, pending or threatened, or of any facts or circumstances that could give rise to any obligation by BancGroup to provide the indemnification required by this section 6.1(h) other than as disclosed in the letters of the directors and executive officers referred to in section 6.1(h) (iii) hereof or described in any schedule to this Agreement and claims arising from any transaction contemplated by this Agreement.
- 6.2 Additional Covenants of Acquired Corporation. Acquired Corporation covenants to and with BancGroup as follows:
 - (a) Operations. (i) Acquired Corporation will conduct its business and the business of each Acquired Corporation Company in a proper and prudent manner and will use its best efforts to maintain its relationships with its depositors, customers and employees. No Acquired Corporation Company will engage in any material transaction outside the ordinary course of business or make any material change in its accounting policies or methods of operation, nor will Acquired Corporation permit the occurrence of any change or event which would render any of the representations and warranties in Article 5 hereof untrue in any material respect at and as of the Effective Date with the same effect as though such representations and warranties had been made at and as of such Effective Date. Acquired Corporation will take no action that would prevent or impede the Merger from qualifying (i) for pooling of interests accounting treatment or (ii) as a reorganization with the meaning of Section 363 of the Code.

- (ii) If requested by BancGroup, Acquired Corporation shall use its best efforts to cause all officers and directors of any Acquired Corporation Company that own any stock of Acquired Corporation and all shamholders of Acquired Corporation who own more than five percent (5%) of Acquired Corporation outstanding shares of common stock, to execute an acknowledgment that such person has no present plan, intention, or binding commitment to sell or otherwise dispose of the BancGroup Common Stock to be received in the Merger within twelve (12) months after the Effective Date.
- (b) Stockholders Meeting; Best Efforts. Acquired Corporation will cooperate with BancGroup in the preparation of the Registration Statement and any regulatory filings and will cause the Stockholders Meeting to be held for the purpose of approving the Merger as soon as practicable after the effective date of the Registration Statement, and will use its best efforts to bring about the transactions contemplated by this Agreement, including stockholder approval of this Agreement, as soon as practicable unless this Agreement is terminated as provided herein.
- (c) Prohibited Negotiations. Until the termination of this Agreement, neither Acquired Corporation nor any of Acquired Corporation's directors or officers (or any person representing any of the foregoing) shall solicit or encourage inquiries or proposals with respect to, furnish any information relating to or participate in any negotiations or discussions concerning, an Acquisition Proposal other than as contemplated by this Agreement. Acquired Corporation will notify BancGroup immediately if any such Acquisition Proposal is received by Acquired Corporation, if any such information is requested from Acquired Corporation, or if any such negotiations or discussions are sought to be initiated with Acquired Corporation, and Acquired Corporation shall instruct Acquired Corporation's officers, directors, agents or affiliates or their subsidiaries to refrain from doing any of the above; provided, however, that Acquired Corporation may communicate information about such an Acquired Corporation (a copy of which shall be provided in advance to BancGroup) that it is required to do so in order to comply with its legal obligations. Acquired Corporation shall immediately cease and cause to be terminated any existing activities, discussions, or negotiations with any Persons other than BancGroup conducted hereto with respect to any of the foregoing.
- (d) Director Recommendation. The members of the Board of Directors of Acquired Corporation agree to support publicly the Merger, provided, however, that nothing contained herein shall be deemed to prohibit any officer or director of Ar quired Corporation from fulfilling his fiduciary duty or from taking any action that is required by Law.
- (e) Shareholder Voting. Acquired Corporation shall on the date of execution of this Agreement obtain and submit to BancGroup an agreement from certain of its shareholders substantially in the form set forth in Exhibit A.
 - (f) Financial Statements. Acquired Corporation shall furnish to BancGroup:
 - (i) As soon as practicable and in any event within 45 days after the end of each quarterly period (other than the last quarterly period) in each fiscal year, consolidated statements of operations of Acquired Corporation for such period and for the period beginning at the commencement of the fiscal year and ending at the end of such quarterly period, and a consolidated statement of financial condition of Acquired Corporation as of the end of such quarterly period, setting forth in each case in comparative form figures for the corresponding periods ending in the preceding fiscal year, subject to changes resulting from year-end adjustments;
 - (ii) Promptly upon receipt thereof, copies of all audit reports submitted to Acquired Corporation by independent auditors in connection with each annual, interim or special audit of the books of Acquired Corporation made by such accountants;
 - (iii) As soon a practicable, copies of all such financial statements and reports as it shall send to its stockholders and of such regular and periodic reports as Acquired Corporation may file with the SEC or any other Agency; and

- (iv) With reasonable promptness, such additional financial data as BancGroup may reasonably request,
- (g) Fiduciary Duties. Prior to the Effective Date, (I) no director or officer (each an "Executive") of any Acquired Corporation Company shall, directly or indirectly, own, manage, operate, join, control, be employed by or participate in the ownership, proposed ownership, management, operation or control of or be connected in any manner with, any business, corporation or partnership which is competitive to the business of any Acquired Corporation Company, (ii) all Executives, at all times, shall satisfy their fiduciary duties to Acquired Corporation and its Subsidiaries, and (iii) such Executives shall not (except as required in the course of his or her employment with any Acquired Corporation Company) communicate or divulge to, or use for the benefit of himself or herself or any other person, firm, association or corporation, without the express written consent of Acquired Corporation, any confidential information which is possessed, owned or used by or licensed by or to any Acquired Corporation Company or confidential information belonging to third parties which any Acquired Corporation Company shall be under obligation to keep secret or which may be communicated to, acquired by or learned of by the Executive in the course of or as a result of his or her employment with any Acquired Corporation Company.
- (h) Certain Practices. At the request of BancGroup, (i) Acquired Corporation shall consult with BancGroup and advise BancGroup through its bank Subsidiary in Orlando, Florida of all of the Lank's loan requests over \$2,000,000 that are not single-family residential loan requests or of any other loan request outside the Bank's normal course of business, and (ii) Acquired Corporation will consult with BancGroup to coordinate, on a basis mutually satisfactory to Acquired Corporation and BancGroup, such business issues as the Acquired Corporation reasonably believes should be brought to the attention of BancGroup. Acquired Corporation and the Bank shall not be required to undertake any of such activities, however, except as such activities may be in compliance with existing Law and Regulations.

ARTICLE 7

MUTUAL COVENANTS AND AGREEMENTS

- 7.1 Best Efforts; Cooperation. Subject to the terms and conditions herein provided, BaneGroup and Acquired Corporation each agrees to use its best efforts promptly to take, or cause to be taken, all actions and do, or cause to be done, all things necessary, proper or advisable under applicable Laws or otherwise, including, without limitation, promptly making required deliveries of stockholder lists and stock transfer reports and attempting to obtain all necessary Consents and waivers and regulatory approvals, to consummate and make effective, as soon as practicable, the transactions contemplated by this Agreement. The officers of each Party to this Agreement shall fully cooperate with officers and employees, accountants, counsel and other representatives of the other Parties not only in fulfilling the duties hereunder of the Party of which they are officers but also in assisting, directly or through direction of employees and other persons under their supervision or control, such as stock transfer agents for the Party, the other Parties requiring information which is reasonably available from such Party.
- 7.2 Press Release. Each Party hereto agrees that, unless approved by the other Parties in advance, such Party will not make any public announcement, issue any press release or other publicity or confirm any statements by any person not a party to this Agreement concerning the transactions contemplated hereby. Notwithstanding the foregoing, each Party hereto reserves the right to make any disclosure if such Party, in its reasonable discretion, deems such disclosure required by Law. In that event, such Party shall provide to the other Party the text of such disclosure sufficiently in advance to enable the other Party to have a reasonable opportunity to comment thereon.
- 7.3 Mutual Disclosure. Each Party hereto agrees to promptly furnish to each other Party hereto its public disclosures and filings not precluded from disclosure by Law including but not limited to call reports, Form 8-K, Form 10-Q and Form 10-K filings, Y-3 applications, reports on Form Y-6, quarterly or special reports to shareholders, Tax returns, Form S-8 registration statements and similar documents.

- 7.4 Access to Properties and Records. Each Party hereto shall afford the officers and authorized representatives of the other Party full access to the Assets, books and records of such Party in order that such other Parties may have full opportunity to make such investigation as they shall desire of the affairs of such Party and shall furnish to such Parties such additional financial and operating data and other information as to its businesses and Assets as shall be from time to time reasonably requested. All such information that may be obtained by any such Party will be held in confidence by such party, will not be disclosed by such Party or any of its representatives except in accordance with this Agreement, and will not be used by such Party for any purpose other than the accomplishment of the Merger as provided herein.
- 7.5 Notice of Adverse Changes. Each Party agrees to give written notice promptly to the other Party upon becoming aware of the occurrence or impending occurrence of any event or circumstance relating to it or any of its Subsidiaries which (i) is reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on it or (ii) would cause or constitute a material breach of any of its representations, warranties or covenants contained herein, and to use its reasonable efforts to prevent or promptly to remedy the same.

ARTICLE 8

CONDITIONS TO OBLIGATIONS OF ALL PARTIES

The obligations of BancGroup and Acquired Corporation to cause the transactions contemplated by this Agreement to be consummated shall be subject to the satisfaction, in the sole discretion of the Party relying upon such conditions, on or before the Effective Date of all the following conditions, except as such Parties may waive such conditions in writing:

- .1 Approval by Shareholders. At the Stockholders Meeting, this Agreement and the matters contemplated by this Agreement shall have been duly approved by the vote of the holders of not less than the requisite number of the issued and outstanding voting securities of Acquired Corporation as is required by applicable Law and Acquired Corporation's articles of incorporation and bylaws.
- 8.2 Regulatory Authority Approval. Orders, Consents and approvals, in form and substance reasonably satisfactory to BancGroup and Acquired Corporation, shall have been entered by the Board of Governors of the Federal Reserve System and other appropriate bank regulatory Agencies (i) granting the authority necessary for the consummation of the transactions contemplated by this Agreement, including the merger of the Bank with Colonial Bank as structured pursuant to section 2.8 hereof and (ii) satisfying all other requirements prescribed by Law. No Order, Consent or approval so obtained which is necessary to consummate the transactions as contemplated hereby shall be conditioned or restricted in a manner which in the reasonable good faith judgment of the Board of Directors of BancGroup would so materially adversely impact the economic benefits of the transaction as contemplated by this Agreement so as to render inadvisable the consummation of the Merger.
- 8.3 Litigation. There shall be no pending or threatened Litigation in any court or any pending or threatened proceeding by any governmental commission, board or Agency, with a view to seeking or in which it is sought to restrain or prohibit consummation of the transactions contemplated by this Agreement or in which it is sought to obtain divestiture, rescission or damages in connection with the transactions contemplated by this Agreement and no investigation by any Agency shall be pending or threatened which might result in any such suit, action or other proceeding.
- 8.4 Registration Statement. The Registration Statement shall be effective under the 1933 Act and no stop order suspending the effectiveness of the Registration Statement shall be in effect; no proceedings for such purpose, or under the proxy rules of the SEC or any bank regulatory authority pursuant to the 1934 Act, and with respect to the transactions con'emplated hereby, shall be pending before or threatened by the SEC or any bank regulatory authority; and all approvals or authorizations for the offer of BancGroup Common Stock shall have been received or obtained pursuant to any applicable state securities Laws, and no stop order or proceeding with respect to the transactions contemplated hereby shall be pending or threatened under any such state Law.

8.5 Tax Opinion. An opinion of Coopers & Lybrand, LLP, shall have been received in form and substance reasonably satisfactory to the Acquired Corporation and BancGroup to the effect that (i) the Merger will constitute a "reorganization" within the meaning of section 368 of the Code; (ii) no gain or loss will be recognized by BancGroup or Acquired Corporation; (iii) no gain or loss will be recognized to the shareholders of Acquired Corporation who receive shares of BancGroup Common Stock except to the extent of any taxable "boot" received by such persons from BancGroup, and except to the extent of any dividends received from Acquired Corporation prior to the Effective Date; (iv) the basis of the BancGroup Common Stock received in the Morger will be equal to the sum of the basis of the shares of Acquired Corporation common stock exchanged in the Merger and the amount of gain, if any, which was recognized by the exchanging Acquired Corporation shareholder, including any portion treated as a dividend, less the value of taxable boot, if any, received by such shareholder in the Merger; (v) the holding period of the BancGroup Common Stock will include the holding period of the shares of Acquired Corporation common stock exchanged therefor if such shares of Acquired Corporation common stock are capital assets in the hands of the exchanging Acquired Corporation shareholder, and (vi) cash received by an Acquired Corporation shareholder in lieu of a fractional share interest of BancGroup Common Stock will be treated as having been received as a distribution in full payment in exchange for the fractional share interest of BancGroup Common Stock which he or she would otherwise be entitled to receive and will qualify as capital gain or loss (assuming the Acquired Corporation common stock is a capital asset in his or her hands as of the Effective Date).

ARTICLE 9

CONDITIONS TO OBLIGATIONS OF ACQUIRED CORPORATION

The obligations of Acquired Corporation to cause the transactions contemplated by this Agreement to be consummated shall be subject to the satisfaction on or before the Effective Date of all the following conditions except as Acquired Corporation may waive such conditions in writing:

- 9.1 Representations, Warranties and Covenants. Notwithstanding any investigation made by or on behalf of Acquired Corporation, all representations and warranties of BancGroup contained in this Agreement shall be true in all material respects on and as of the Effective Date as if such representations and warranties were made on and as of such Effective Pate, and BancGroup shall have performed in all material respects all agreements and covenants required by this Agreement to be performed by it on or prior to the Effective Date.
- 9.2 Adverse Changes. There shall have been no changes after the date of the most recent balance sheet provided under section 4.3(a) (i) hereof in the results of operations (as compared with the corresponding period of the prior fiscal year), Assets, Liabilities, financial condition or effairs of BaneGroup which in their total effect constitute a Material Adverse Effect, nor shall there have been any material changes in the Laws governing the business of BaneGroup or which would impair the rights of Acquired Corporation or its shareholders pursuant to this Agreement.
- 9.3 Closing Certificate. In addition to any other deliveries required to be delivered hereunder, Acquired Corporation shall have received a certificate from the President or E. Vice President and from the Secretary or Assistant Secretary of BancGroup dated as of the Closing certifying that:
 - (a) the Board of Directors of BancGroup has duly adopted resolutions approving the substantive terms of this Agreement and authorizing the consummation of the transactions contemplated by this Agreement and such resolutions have not been amended or modified and remain in full force and effect;
 - (b) each person executing this Agreement on behalf of BancGroup is an officer of BancGroup holding the office or offices specified therein and the signature of each person set forth on such certificate is his or her genuine signature;
 - (c) the certificate of incorporation and bylaws of BancGroup referenced in section 4.4 hereof remain in full force and effect;
 - (d) such persons have no knowledge of a basis for any material claim, in any court or before any Agency or arbitration or otherwise against, by or affecting BancGroup or the business, prospects,

condition (financial or otherwise), or Assets of BaneGroup or which would prevent the performance of this Agreement or the transactions contemplated by this Agreement or declare the same unlawful or cause the resolution thereof.

- (c) to such persons' knowledge, the Proxy Statement delivered to Acquired Corporation's shareholders, or any amendments or revisions thereto so delivered, as of the date thereof, did not contain or incorporate by reference any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading 12 light of the circumstances under which they were made (it being understood that such persons need not express a statement as to information concerning or provided by Acquired Corporation for inclusion it such Proxy Statement); and
 - (f) the conditions set forth in this Article 9 insofar as they relate to BancGroup have been satisfied.
- 9.4 Opinion of Counsel. Acquired Corporation shall have received an opinion of Miller, Hamilton, Snider & Odom, L.L.C., counsel to BancGroup, dated as of the Closing, substantially in the form set forth in Exhibit B hereto.
- 9.5 Fairness Opinion. Acquired Corporation shall have received prior to the mailing of the Proxy Statement from The Carson Mediin Company a letter setting forth its opinion that the Merger Consideration to be received by the shareholders of Acquired Corporation under the terms of this Agreement is fair to them from a financial point of view, and such opinion shall not have been withdrawn as of the Effective Date.
- 9.6 NYSE Listing. The shares of BancGroup Common Stock to be issued under this Agreement shall have been approved for listing on the NYSE.
- 9.7 Other Matters. There shall have been furnished to counsel for Acquired Corporation certified copies of such corporate records of BancGroup and copies of such other documents as such counsel may reasonably have requested for such purpose.
- 9.8 Material Events. There shall have been no determination by the board of directors of Acquired Corporation that the transactions contemplated by this Agreement have become impractical because of any state of war, declaration of a banking moratorium in the United States or a general suspension of trading on the NYSE or any other exeb or on, which BancGroup Common Stock may be traded.

ARTICLE 10

CONDITIONS TO OBLIGATIONS OF BANCGROUP

The obligations of BancGroup to cause the transactions contemplated by this Agreement to be consummated shall be subject to the satisfaction on or before the Effective Date of all of the following conditions except as BancGroup may waive such conditions in writing:

- 10.1 Representations, Warranties and Covenants. Notwithstanding any investigation made by or on behalf of BancGroup, all representations and warranties of Acquired Corporation contained in this Agreement shall be true in all material respects on and as of the Effective Date as if such representations and warranties were made on and as of the Effective Date, and Acquired Corporation shall have performed in all material respects all agreements and covenants required by this Agreement to be performed by it on or prior to the Effective Date.
- 10.2 Adverse Changes. There shall have been no changes after the date of the most recent balance sheet provided under section 5.4(a) (i) hereof in the results of operation: (as compared with the corresponding period of the prior fiscal year), Assets, Liabilities, financial condition or affairs of Acquired Corporation which constitute a Material Adverse Effect, nor shall there have been any material changes in the Laws governing the business of Acquired Corporation which would impair BancGroup's rights to resum to this Agreement.

- 10.3 Closing Certificate. In addition to any other deliveries required to be delivered hereunder, BancGre in shall have received a certificate from Acquired Corporation executed by the President or Vice President and from the Secretary or Assistant Secretary of Acquired Corporation dated as of the Closing certifying that:
 - (a) the Board of Directors of Acquired Corporation has duly adopted resolutions approving the substantive terms of this Agreement and authorizing the consummation of the transactions contemplated by this Agreement and such resolutions have not been amended or modified and remain in full force and effect:
 - (b) the shareholders of Acquired Corporation have duly adopted resolutions approving the substantive terms of the Merger and the transactions contemplated thereby and such resolutions have not been amended or modified and remain in full force and effect:
 - (c) each person executing this Agreement on behalf of Acquired Corporation is an officer of Acquired Corporation holding the office or offices specified therein and the signature of each person set forth on such certificate is his or her genuine signature;
 - (d) the articles of incorporation and bylaws of Acquired Corporation and the Bank referenced in section 5.8 hereof remain in full force and effect and have not been amended or modified since the date hereof:
 - (e) to such persons' knowledge, the Proxy Statement delivered to Acquired Corporation's share-holders, or any amendments or revisions thereto so delivered, as of the date thereof, did not contain or incorporate by reference any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances under which they were made (it being understood that such persons need only express a statement as to information concerning or provided by Acquired Corporation for inclusion in such Proxy Statement); and
 - (f) the conditions set forth in this Article 10 insofar as they relate to Acquired Corporation have been satisfied.
- 10.4 Opinion of Counsel. BancGroup shall have received an opinion of Bush Ross Gardner Warren & Rudy, P.A., counsel to Acquired Corporation, dated as of the Closing, substantially as set forth in Exhibit C hereto.
- 10.5 Controlling Shareholders. Each shareholder of Acquired Corporation who may be an "affiliate" of Acquired Corporation, within the meaning of Rule 145 of the general rules and regulations under the 1933 Act shall have executed and delivered an agreement satisfactory to BancGroup to the effect that such person shall not make a "distribution" (within the meaning of Rule 145) of the Common Stock which he receives upon the Effective Date and that such Common Stock will be held subject to all applicable provisions of the 1933 Act and the rules and regulations of the SEC thereunder, and the undersigned will not sell or otherwise reduce risk relative to any shares of BancGroup Common Stock received in the Merger until financial results concerning at least 30 days of post-Merger combined operations have been published by BancGroup within the meaning of Section 201.01 of the SEC's Codification of Financial Reporting Policies. Acquired Corporation recognizes and acknowledges that BancGroup Common Stock issued to such persons may bear a legend, in form reasonably acceptable to the Acquired Corporation, evidencing the agreement described above.
- 10.6 Other Matters. There shall have been furnished to counsel for BancGroup certified copies of such corporate records of Acquired Corporation and copies of such other documents as such counsel may reasonably have requested for such purpose.
- 10.7 Dissenters. The number of shares as to which shareholders of Acquired Corporation shall have exercised dissenters rights of appraisal under section 3.6 does not exceed 10 percent of the outstanding shares of common stock of Acquired Corporation.

10.8 Material Events. There shall have been no determination by the board of directors of BancGroup that the transactions contemplated by this Agreement have become impractical because of any state of war, declaration of a banking moratorium in the United States or general suspension of trading on the NYSE or any exchange on which BancGroup Common Stock may be traded.

10.9 Severance Agreement. A severance agreement, in form and substance reasonably satisfactory to BancGroup and Richard H. Eatman, shall have been executed between BancGroup and Richard H. Eatman.

10.10 Pooling of Interest. BancGroup shall have received the written opinion of Coopers & Lybrand, L.L.P., that the Merger will qualify for the pooling of interests method of accounting under generally accepted accounting principles.

ARTICLE 11

TERMINATION OF REPRESENTATIONS AND WARRANTIES

All representations and warranties provided in Articles 4 and 5 of this Agreement or in any closing certificate pursuant to Articles 9 and 10 shall terminate and be extinguished at and shall not survive the Effective Date. All covenants, agreements and undertakings required by this Agreement to be performed by any Party hereto following the Effective Date shall survive such Effective Date and be binding upon such Party. If the Merger is not consummated, all representations, warranties, obligations, covenants, or agreements hereunder or in any certificate delivered hereunder relating to the transaction which is not consummated shall be deemed to be terminated or extinguished, except that Sections 7.2, 7.4, Article 11, Article 15 and any applicable definitions of Article 14, shall survive. Items disclosed in the Exhibits and Schedules attached hereto are incorporated into this Agreement and form a part of the representations, warranties, covenants or agreements to which they relate. Information provided in such Exhibits and Schedules is provided only in response to the specific section of this Agreement which calls for such information.

ARTICLE 12

NOTICES

All notices, requests, demands, consents, approvals and other communications under this Agreement shall be in writing and shall be given to the recipient party by hand delivery; or by sending a copy thereof by first class or express mail, postage prepaid, by telegram (with messenger service specified), or by courier service (with charges prepaid), in each case to the address indicated below or to such other address as the recipient shall have provided in accordance with the terms hereof; or by sending a copy thereof by whatever telecopier service the recipient shall have designated below (or by subsequent notice provided in accordance with the terms hereof). If the notice is sent by mail, telegraph or courier services, it shall be deemed to have been given to the recipient when deposited in the United States mail or with a telegraph office or courier service for delivery to that party; or if by telecopier, when the sending party is in receipt of documentary evidence that the transmission has been successfully completed. Whenever the furnishing of notice is required,

the same may be waived by the party entitled to receive such notice. Until further notice is provided, the address and telecopier number of each party is as stated below:

/is to BancGroup: One Commerce Street, Suite F00

Montgomery, Alabama 33602 Attention: W. Flake Oakley, IV Telecopier No. (334) 240-601

Copy to: Michael D. Waters, Esq.

Miller, Hamilton, Snider & Odom, L.L.C.

One Commerce Street, Suite 802 Montgomery, Alabama 36104 Telecopier No. (334) 265-4533

As to Acquired Corporation:

510 Vonderburg Drive Brandon, Florida 33511

Attention: Richard H. Eatman, President

Telecopier No. (813) 651-1629

Copy to: Jeromy P. Ross, Esq.

Bush Ross Gardner Warren & Rudy, P.A.

220 South Franklin Street Tampa, Florida 33602

Telecopier No. (813) 223-9620

ARTICLE 13

AMENDMENT OR TERMINATION

- 13.1 Amendment. This Agreement may be amended by the mutual consent of BancGroup and Acquired Corporation before or after approval of the transactions contemplated herein by the shareholders of Acquired Corporation.
- 13.2 Termination. This Agreement may be terminated at any time prior to or on the Effective Date whether before or after action thereon by the shareholders of Acquired Corporation, as follows:
 - (a) by the mutual consent of the respective boards of directors of Acquired Corporation and BancGroup;
 - (b) by the board of directors of either Party (provided that the terminating Party is not then in material breach of any representation, warranty, covenant, or other agreement contained in this Agreement) in the event of a breach by the other Party of any representation or warranty contained in this Agreement which cannot be or has not been cured within thirty (30) days after the giving of written notice to the breaching Party of such breach and which breach would provide the non-breaching Party the ability to refuse to consummate the Me. ger for failure to satisfy the condition set forth in section 10.1 of this Agreement in the case of BancGroup and section 9.1 of this Agreement in the case of Acquired Corporation;
 - (c) by the board of directors of either Party (provided that the terminating Party is not then in material b each of any representation, warranty, covenant, or other agreement contained in this Agreement) in the event of a material breach by the other Party of any covenant or agreement contained in this Agreement which cannot be or has not been cured within thirty (30) days after the giving of written notice to the breaching Party of such breach, or if any of the conditions to the obligations of such Party contained in this Agreement in Article 9 as to Acquired Corporation or Article 10 as to BancGroup shall not have been satisfied in full; or
 - (d) by the board of directors of either BancGroup or Acquired Corporation if all transactions contemplated by this Agreement shall not have been consummated on or prior to June 30, 1997, if the

failure to consummate the transactions provided for in this Agreement on or before such date is not caused by any breach of this Agreement by the Party electing to terminate pursuant to this section 13.2(d).

13.3 Damages. In the event of termination pursuant to section 13.2, Acquired Corporation and BancGroup shall not be liable for damages for any breach of warranty or representation contained in this Agreement made in good faith, and, in that case, the expenses incurred shall be borne as set forth in section 15.1 hereof.

ARTICLE 14

DEFINITIONS

(a) The following terms, which are capitalized in this Agreement, shall have the meanings set forth

below for the purpose of this Agreement: Acquired Corporation Fort Brooke Bancorporation, a Florida corporation. Acquired Corporation Company Shall mean Acquired Corporation, the Bank, any Subsidiary of Acquired Corporation or the Bank, or any person or entity acquired as a Subsidiary of Acquired Corporation or the Bank in the future and owned by Acquired Corporation or the Bank at the Effective Date. Acquired Corporation Options... Options respecting the issuance of 60,410 shares of Acquired Corporation common stock pursuant to Acquired Corporation's stock option pleas. Acquired Corporation Stock Shares of common stock, par value \$.001 per share, of Acquired Corporation. Acquisition Proposal Shall mean, with respect to a Party, any tender offer or exchange offer or any proposal for a merger, acquisition of all of the stock or assets of, or other business combination involving such Party or any of its Subsidiaries or the acquisition of a substantial equity interest in, or a substantial portion of the assets of, such Party or any of its Subsidiaries. Shall mean, collectively, the Federal Trade Commission, the United States Department of Justice, the Board of Governors of the Federal Reserve System, the Federal Deposit Insurance Corporation, the Office of Thrift Supervision, all state regulatory agencies having jurisdiction over the Parties and their respective Subsidiaries, HUD, the VA, the FHA, the GNMA, the FNMA, the FHLMC, the NYSE, and the SEC. Agreement Shall mean this Agreement and Plan of Merger and the Exhibits and Schedules delivered pursuant hereto and incorporated herein by poierence. Of a Person shall mean all of the assets, properties, businesses and rights Assets of such Person of every kind, nature, character and description, whether real, personal or mixed, tangible or intangible, accrued or contingent, or otherwise relating to or utilized in such Person's business, directly or indirectly, in whole or in part, whether or not carried on the books and records of such Person, and whether or not owned in the name of such Person or any Affiliate of such Person and wherever located. The Colonial BancGroup, Inc., a Delaware corporation with its principal BancGroup..... offices in Montgomery, Alabama.

Bank	Fort Brooke Bank, a Florida state bank.
Closing	The closing of the transactions contemplated hereby as described in section 2.7 of this Agreement.
Code	The Internal Revenue Code of 1986, as amended.
Colonial Bank	BancGroup's Subsidiary bank located in Orlando, Florida.
Common Stock	BaneGroup's Common Stock authorized and defined in the restated certificate of incorporation of BaneGroup, as amended.
Consent	Any consent, approval, authorization, clearance, exemption, waiver, or similar affirmation by any Person pursuant to any Contract, Law, Order, or Permit.
Contract	Any written or oral agreement, arrangement, authorization, commitment, contract, indenture, instrument, lease, obligation, plan, practice, restriction, understanding or undertaking of any kind or character, or other document to which any Person is a party or that is binding on any Person or its capital stock, Assets or business.
Default	Shall mean (i) any breach or violation of or default under any Contract, Order or Permit, (ii) any occurrence of any event that with the passage of time or the giving of notice or both would constitute a breach or violation of or default under any Contract, Order or Permit, or (iii) any occurrence of any event that with or without the passage of time or the giving of notice would give rise to a right to terminate or revoke, change the current terms of, or renegotiate, or to accelerate, increase, or impose any Liability under, any Contract, Order or Permit.
DGCL	The Delaware General Corporation Law.
Effective Date	Means the date and time at which the Merger becomes effective as defined in section 2.7 hereof.
Environmental Laws	Means the laws, regulations and governmental requirements referred to in section 5.23 hereof.
ERISA	The Employee Retirement Income Security Act of 1974, as amended.
Exchange Ratio	The ratio obtained by dividing \$31.50 by the Market Value, as set forth in section 3.1(b).
Exercise Price	The exercise price of Acquired Corporation Options as set forth in section 3.1(b) (ii).
Exhibits	A through C, inclusive, shall mean the Exhibits so marked, copies of which are attached to this Agreement. Such Exhibits are hereby incorporated by reference herein and made a part hereof, and may be referred to in this Agreement and any other related instrument or document without being attached hereto.
FBCA	The Florida Business Corporation Act
Knowledge	Means the actual knowledge of the Chairman, President, Chief Financial Officer, Chief Accounting Officer, Chief Credit Officer, General Counsel or any Senior or Executive Vice President of BancGroup, in the case of knowledge of BancGroup, or of Acquired Corporation and the Bank, in the case of knowledge of Acquired Corporation.

•	
Law	Any code, law, ordinance, regulation, reporting or licensing requirement, rule, or statute applicable to a Person or its Assets, Liabilities or business, including those promulgated, interpreted or enforced by any Agency.
Linbility	Any direct or indirect, primary or accordary, liability, indebtedness, obligation, penalty, cost or expense (including costs of investigation, collection and defense), deficiency, guaranty or endorsement of or by any Person (other than endorsements of notes hills, checks, and drafts presented for collection or deposit in the ordinary course of business) of any type, whether accrued, absolute or contingent, liquidated or unliquidated, matured or unmatured, or otherwise.
Lien	Any conditional sale agreement, confault of title, easement, encroachment, encumbrance, hypothecation, infringement, lien, mortgage, pledge, reservation, restriction, security interest, title retention or other security arrangement, or any adverse right or interest, charge, or claim of any nature whatsoever of, on, or with respect to any property or property interest, other than (i) Liens for current property Taxes not yet due and payable, (ii) for depository institution Subsidiaries of a Party, pledges to secure deposits and other Liens incurred in the ordinary course of the banking business, and (iii) Liens in the form of easements and restrictive covenants on real property which do not create a Material Adverse Effect with respect to the use of such property by the current owner thereof.
Litigation	Any action, arbitration, complaint, criminal prosecution, governmental or other examination or investigation, hearing, inquiry, administrative or other proceeding relating to or affecting a Party, its business, its Assets (including Contracts related to it), or the transactions contemplated by this Agreement.
Loss	Any and all direct or indirect payments, obligations, recoveries, deficiencies, fines, penalties, interest, assessments, losses, diminution in the value of Assets, damages, punitive, exemplary or consequential damages (including, but not limited to, lost income and profits and interruptions of business), liabilities, costs, expenses (including without limitation, reasonable attorneys' fees and expenses, and consultant's fees and other costs of defense or investigation), and interest on any amount payable to a third party as a result of the foregoing.
Material	For purposes of this Agreement shall be determined in light of the facts and circumstances of the matter in question; provided that any specific monetary amount stated in this Agreement shall determine materiality in that instance.
Material Adverse Effect	On a Party shall mean an event, change or occurrence which has a material adverse impact on (i) the financial position, Assets, business, or results of operations of such Party and its Subsidiaries, taken as a whole, or (ii) the ability of such Party to perform its obligations under this Agreement or to consummate the Merger or the other transactions contemplated by this Agreement, provided that "material adverse impact" shall not be deemed to include the impact of (x) changes in banking and similar laws of general applicability or interpretations thereof by courts or governmental authorities, (y) changes in generally accepted accounting principles or regulatory accounting principles generally applicable to banks and their holding companies, and (z) the Merger

and compliance with the provisions of this Agreement on the operating performance of the Parties. The merger of Acquired Corporation : th BancGroup as contemplated Morger in this Agreement. The distribution of BancGroup Common Stock and cash for each share of Acquired Corporation Stock as provided in section 3.1(a) hercof. NYSE...... The New York Stock Exchange. Order Any administrative decision or award, decree, injunction, judgment, order, quasi-judicial decision or award, ruling, or writ of any federal, state, local or foreign or other court, arbitrator, mediator, tribunal, administrative agency or Agency. Shall mean Acquired Corporation or BancGroup, and "Parties" shall mean both Acquired Corporation and BancGroup. Any federal, state, local, and foreign governmental approval, authoriza-Permit..... tion, certificate, easement, filing, franchise, license, notice, permit, or right to which any Person is a party or that is or may be binding upon or inure to the benefit of any Person or its securities, Assets or business. A natural person or any legal, commercial or governmental entity, such as, but not limited to, a corporation, general partnership, joint venture, limited partnership, limited liability company, trust, business association, group acting in concert, or any person acting in a representative capacity. The proxy statement used by Acquired Corporation to solicit the ap-Proxy Statement proval of its stockholders of the transactions contemplated by this Agreement, which shall include the prospectus of BancC. up relating to the issuance of the BancGroup Common Stock to the shareholders of Acquired Corporation. The registration statement on Form S-4, or such other appropriate form, Registration Statement to be filed with the SEC by BancGroup, and which has been agreed to by Acquired Corporation, to register the shares of BancGroup Common Stock offered to stockholders of the Bank pursuant to his Agreement, including the Proxy Statement. Resulting Corporation BancGroup, as the surviving corporation resulting from the Merger. SEC United States Securities and Exchange Commission. Stockholders Meeting The special meeting of stockholders of Acquired Corporation called to approve the transactions contemplated by this Agreement. Shall mean all those corporations, banks, associations, or other entities of Subsidiaries which the entity in question owns or controls 5% or more of the outstanding equity securities either directly or through an unbroken chain of entities as to each of which 5% or more of the outstanding equity securities is owned directly or indirectly by its parent; provided, however, there shall not be included any such entity acquired through foreclosure or any such entity the equity securities of which are owned or controlled in a fiduciary capacity. Tax or Taxes Means any federal, state, county, local, foreign, and other taxes, assessments, charges, fares, and impositions, including interest and penalties

thereon or with respect thereto.

ARTICLE 15

MISCELLANEOUS

- 15.1 Expenses. Each Party hereto shall bear its own legal, auditing, trustee, investment banking, regulatory and other expenses in connection with this Agreement and the transactions contemplated hereby.
- 15.2 Benefit. This Agreement shall inure to the benefit of and be binding upon Acquired Corporation and BaneGroup, and their respective successors. This Agreement shall not be assignable by any Party without the prior written consent of the other Party.
- 15.3 Governing Law. Except to the extent that the laws of the State of Florida apply to the Merger, this Agreement shall be governed by, and construed in accordance with the Laws of the State of Delaware without regard to any conflict of Laws.
- 15.4 Counterparts. This Agreement may be executed in counterparts, each of which shall be deemed to constitute an original. Each such counterpart shall become effective when one counterpart has been signed by each Party the sto.
- 15.5 Headings. The headings of the various articles and sections of this Agreement are for convenience of reference only and shall not be deemed a part of this Agreement or considered in construing the provisions thereof.
- 15.6 Severability. Any term or provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining terms and provisions thereof or affecting the validity or enforceability of such provision in any other jurisdiction, and if any term or provision of this Agreement is held by any court of competent jurisdiction to be void, voidable, invalid or unenforceable in any given circumstance or situation, then all other terms and provisions, being severable, shall remain in full force and effect in such circumstance or situation and the term or provision shall remain valid and in effect in any other circumstances or situation.
- 15.7 Construction. Use of the masculine pronoun herein shall be deemed to refer to the feminine and neuter genders and the use of singular references shall be deemed to include the plural and vice versa, as appropriate. No inference in favor of or against any Party shall be drawn from the fact that such Party or such Party's counsel has drafted any portion of this Agreement.
- 15.8 Return of Information. In the event of termination of this Agreement prior to the Effective Date, each Party shall return to the other, without retaining copies thereof, all confidential or non-public documents, work papers and other materials obtained from the other Party in connection with the transactions contemplated in this Agreement and shall keep such information confidential, not disclose such information to any other person or entity, and not use such information in connection with its business.
- 15.9 Equitable Remedies. The parties hereto agree that, in the event of a breach of this Agreement by either Party, the other Party may be without an adequate remedy at law owing to the unique nature of the contemplated transactions. In recognition thereof, in addition to (and not in lieu of) any remedies at law that may be available to the non-breaching Party, the non-breaching Party shall be entitled to obtain equitable relief, including the remedies of specific performance and injunction, in the event of a breach of this Agreement by the other Party, and no attempt on the part of the non-breaching Party to obtain such equitable relief shall be deemed to constitute an election of remedies by the non-breaching Party that would preclude the non-breaching Party from obtaining any remedies at law to which it would otherwise be entitled.
- 15.10 Attorneys' Fees. If any Party hereto shall bring an action at law or in equity to enforce its rights under this Agreement (including an action based upon a misrepresentation or the breach of any warranty, covenant, agreement or obligation contained herein), the prevailing Party in such action shall be entitled to